ERASMUS UNIVERSITY ROTTERDAM ERASMUS SCHOOL OF ECONOMICS Bachelor Thesis Economics & Business Specialization: Marketing

### Examining the Influence of Scarcity and Popularity Cues on Sneaker Purchase Behaviour in the Netherlands: Self vs. Other Consumption Targets

Author:Thalise LacléStudent number:535549Thesis supervisor:Fleur PrinsSecond reader:Tor Kadioglu, CFinish date:20 July 2024

The views stated in this thesis are those of the author and not necessarily those of the supervisor, second reader, Erasmus School of Economics or Erasmus University Rotterdam.

#### **Executive Summary**

This study investigates the effects of scarcity and popularity cues on consumers' purchase intentions for different consumption targets in the Netherlands, particularly in the context of sneakers. It focused on the mediators perceived product uniqueness, perceived consumption risk, and perceived value, as well as the moderator's personal need for uniqueness and self-other overlap.

It builds further on similar research by Wu and Lee, 2016 by first addressing a literature gap by focusing on a high-involvement product category in the fashion industry (sneakers). Secondly, by researching in a culturally different, non-American context (the Netherlands). Thirdly, it incorporates the personal need for uniqueness as a moderating variable, accounting for individual differences in the need for uniqueness instead of assuming a homogenous need for uniqueness. These contributions are valuable for academics and researchers in marketing, psychology, and consumer behaviour.

How do scarcity (limited edition) and popularity (bestseller) cues influence the purchase intentions of consumers in the Netherlands differently when purchasing sneakers for oneself versus as a gift for others?

I will also answer the following empirical sub-questions:

- 1. Does perceived product uniqueness and perceived product value mediate the effect of scarcity cues on purchase intention for self-purchases?
- 2. Does perceived consumption risk and perceived product value mediate the effect of popularity cues on purchase intention?
- 3. Does the personal need for uniqueness moderate the effect of scarcity cues (vs. popularity cues) on purchase intention for self-purchases?
- 4. Does the degree of self-other overlap moderate the effect of cue type on purchase intention for other-purchases?

To answer the questions mentioned above adequately with my research, I will first collect answers to the following theoretical sub-questions:

- 1. How does purchase intention serve as a measure of consumers' likelihood to buy a product?
- 2. What are scarcity-based promotions (limited edition), and how are they used in marketing strategies to influence purchase intention?
- 3. What are popularity-based promotions (bestsellers), and how are they used in marketing strategies to influence purchase intention?
- 4. How would the consumption target influence the effect of scarcity and popularity cues on purchase intention differently, and what is the role of perceived product uniqueness and consumption risk in this?
- 5. What does the "Limited Edition for Me and Bestseller for You effect" of Wu et al. (2016) entail?

- 6. How would personal need for uniqueness influence the effect of scarcity and popularity cues on purchase intention?
- 7. How would the degree of self-other overlap influence the effect of scarcity and popularity cues on purchase intention?

The literature review describes that purchase intention serves as an effective measure of consumers' likelihood to buy a product. Scarcity-based promotions are marketing strategies that emphasise a product's limited availability to create a sense of urgency and exclusivity. Popularity-based promotions are marketing strategies that show the popularity of a particular product in terms of the number of times it has been purchased. The literature review then describes how the consumption target can influence the effect of scarcity and popularity cues on purchase intentions and explains in which way perceived product uniqueness and consumption risk are essential mediators to that effect. Scarcity cues, which are found to enhance perceived product uniqueness, are more effective in increasing purchase intentions for self-purchases. Popularity cues reduce consumption risk for the buyer through social validation, thus increasing purchase intentions for other-purchases. The literature review further finds that consumers with a high need for uniqueness respond more to scarcity cues, increasing their purchase intentions for limited edition products for self-purchases. Finally, it is found that low self-other overlap decreases the effectiveness of popularity cues for other-purchases.

To answer the research questions, an online survey was conducted with a sample of 162 sneaker consumers in the Netherlands. The survey used a  $2 \times 2$  between-subjects experimental design, assigning participants to one of four conditions: limited edition for self-purchase, limited edition for other-purchase, bestseller for self-purchase, and bestseller for other-purchase. Various statistical analyses, including independent t-tests, mediation, and moderation analyses, were performed using Stata and SPSS software.

The main research question is answered as follows: Depending on the consumption target, namely, if a consumer is purchasing sneakers for oneself or purchasing sneakers as a gift for a friend, the effect of scarcity (limited edition) and popularity (bestseller) cues on purchase intention has a different relative performance. For self-purchases, scarcity (limited edition) cues increase purchase intentions more than popularity (bestseller) cues. Opposite to this result, for other-purchases, popularity (bestseller) cues increase purchase intentions more than scarcity (limited edition) cues.

Based on the results of this study, the following hypotheses are accepted/rejected:

Hypothesis	Result
H1. For self-purchases, scarcity (vs. popularity) cues lead to increased purchase	Accepted
intentions.	
H2. For self-purchases, the serial mediation effect of perceived product uniqueness	Rejected
and perceived product value explains the positive impact of scarcity (vs. popularity)	
cues on purchase intentions.	

H3. For other-purchases, popularity (vs. scarcity) cues lead to increased purchase	Accepted
intentions.	(10%
	significance)
H4. For other-purchases, the serial mediation effect of perceived consumption risk	Accepted
and perceived product value explains the positive impact of popularity (vs. scarcity)	(10%
cues on purchase intentions.	significance)
H5. For self-purchases, personal need for uniqueness moderates the effect of cue type	Accepted
on purchase intention such that scarcity cues (vs popularity cues) increase purchase	
intention only when need for uniqueness is high.	
H6. For other-purchases, the degree of self-other overlap moderates the effect of cue	Accepted
type on purchase intentions, such that popularity cues (vs. scarcity cues) lead to	
increased purchase intentions only when self-other overlap is low.	

Based on the findings, the key managerial recommendations are:

- For self-purchases, scarcity cues should be used in the marketing strategy.
- For other-purchases, popularity cues should be used in the marketing strategy.
- Use personalised advertisement based on the consumption target and personal need for uniqueness of online customers if consumer data is available.

Future research recommendations are focused on enhancing the generalisability of the results, most importantly, by conducting field research, analyzing other markets, and controlling for price levels.

**Keywords:** Scarcity, Popularity, Cues, Self-purchase versus other purchase, Product uniqueness, Consumption risk, Personal need for uniqueness, and E-commerce.

### Table of Contents

A	Abstractvii			
1	l Introduction1			
	1.1	Background1		
	1.2	Problem Statement: Central Research Question and Sub-Questions		
	1.3	Research Relevance		
	1.3.1	Academic Relevance		
	1.3.2	Managerial Relevance4		
	1.3.3	Societal Relevance		
	1.4	Research Structure		
2	Liter	ature Study		
	2.1	How does purchase intention serve as a measure of consumers' likelihood to buy a product?		
	2.2	What are scarcity-based promotions (limited edition), and how are they used in marketing		
	strategi	es to influence purchase intention?		
	2.3	What are popularity-based promotions (bestseller) and how are they used in marketing		
	strategi	es to influence purchase intention?		
	2.4	What does the "Limited Edition for Me and Bestseller for You effect" of Wu et al. (2016)		
	entail?			
	2.5	How would the personal need for uniqueness influence the effect of scarcity and popularity		
	cues on	purchase intention?		
	2.6	How would the degree of self-other overlap influence the effect of scarcity and popularity cu	ıes	
	on purc	hase intention?		
	2.7	Conceptual Model		
3	Rese	arch Methodology12		
	3.1	Data Collection: Qualitative or Quantitative research		
	3.2	Survey Overview		
	3.3	Research Sample		
	3.4	Data Analysis		
4	Rese	arch Outcome16		
	4.1	Survey Reliability		
	4.2	Survey Outcome Demographics		
	4.3	Hypothesis 1		
	4.4	Hypothesis 2		
	4.4.1	Direct effects		
	4.4.2	Total and Direct effects		
	4.4.3	Indirect effects bootstrapped at 95 percent		
	4.5	Hypothesis 3		

	4.6	Hypothesis 4	21
	4.6.1	Direct effects	21
	4.6.2	Total and direct effects comparison	22
	4.6.3	Indirect effects bootstrapped at 95 percent	22
	4.6.4	Indirect effects bootstrapped at 90 percent	23
	4.7	Hypothesis 5	24
	4.8	Hypothesis 6	25
	4.9	Summary of Key Findings	27
5	Conc	lusion & Recommendations	28
	5.1	Key Findings Literature	28
	5.2	Key Findings Research	29
	5.3	Conclusion	30
	5.4	Research Limitations	32
	5.5	Recommendation for Future Research	33
	5.6	Managerial Recommendation	33
	5.7	Reflection	34
A	ppendix		35
	Append	lix A- References	35
	Append	lix B – Tables and Figures	39
Appendix C - Survey Questions			46
	Append	lix D – Raw Organized Survey Data	54
	Append	lix E – Statistical test results/output	30

#### Abstract

This study investigates the effects of scarcity and popularity cues on consumers' purchase intentions for different consumption targets in the Netherlands, specifically for sneakers. It also incorporates the mediators perceived product uniqueness, perceived consumption risk, and perceived value, as well as the moderators personal need for uniqueness and self-other overlap. The main research question is whether scarcity (limited edition) and popularity (bestseller) cues influence the purchase intentions of consumers in the Netherlands differently when purchasing sneakers for oneself versus as a gift for others.

An online survey with a sample of 162 sneaker consumers utilised a  $2 \times 2$  between-subjects experimental design. It assessed purchase intentions under four conditions: limited edition for self-purchase, limited edition for other-purchase, bestseller for self-purchase, and bestseller for other-purchase. Statistical analyses, including t-tests, mediation, and moderation analyses, were conducted using Stata and SPSS.

The results indicate that scarcity cues enhance purchase intentions more for self-purchases, while popularity cues are more effective for other-purchases. For self-purchases, these effects are mediated by perceived product uniqueness but not by perceived product value. For other-purchases, these effects are serially mediated by perceived consumption risk and perceived product value. Additionally, it is found that personal need for uniqueness and self-other overlap significantly moderate the effects of scarcity and popularity cues depending on the consumption target on purchase intentions. The study finally provides managerial recommendations for targeted marketing strategies and suggests directions for future research to enhance the generalizability and applicability of these findings.

#### 1 Introduction

#### 1.1 Background

Imagine browsing the web for sneakers and coming across an advertisement for sneakers. One of the ads shows, "These sneakers are a limited edition." The other ad says, "These sneakers are a bestseller among consumers." How would you react to these promotions? Does the purpose of your purchase (e.g., buying for yourself versus as a gift for someone else) affect your response and choice? Would you display a more (or less) favourable reaction toward the sneakers promoted by a scarcity cue (limited edition) or a popularity cue (bestseller)? These questions are relevant for marketers to create effective promotional campaigns tailored to the preferences of various consumer groups in the Netherlands.

Sneaker marketing has become a popular niche, prominently featured on billboards, magazines, TV, and social media. Originally designed to enhance athletic performance, sneakers are now an essential part of daily life and a fashion staple. They are a globally consumed commodity with significant value for consumers, offering a sense of belonging and identity. (Choi, 2017). Last month, Adidas expected to generate an operating profit of around  $\in$ 700 million, a substantial increase from the previous target of  $\notin$ 500 million. (Bloomberg, 2024) A prominent role in this increase is played by Adidas Samba's rise in popularity, mainly driven by young people on social media who desire to fit in the trend of these popular shoes but at the same time want to stand out with Adidas' unique limited-edition offerings in many colours. In this thesis, the relationship between scarcity and popularity cues on purchase intentions for sneakers of consumers in the Netherlands will be researched. Specifically, the difference between purchases for oneself or someone else, called the consumption target.

In retail, scarcity and popularity cues are common tactics to influence customers. Scarcity cues focus on limited availability, like using the term "limited edition" to increase attractiveness for a consumer. In contrast, popularity cues highlight how well-liked a product is, for instance, by calling it a "bestseller", also to make it attractive to the consumer. However, these strategies are incompatible because they are based on conflicting theories of providing uniqueness or the opposite, namely generality. (Deval et al. 2013; Steinhart et al. 2014) This is why they are rarely used together (Gierl, Plantsch, & Schweidler, 2008). Psychological research shows that specific purchasing goals, such as self-expression and individuation, are more relevant to decision-making regarding oneself. (Abele & Wojciszke, 2007) It is interesting to assess how the consumption target influences the effect of scarcity and popularity cues on purchase intentions. The sneaker market, valued at over \$80.19 billion and projected to grow to \$98.14 billion by 2028 (Statista, n.d.), provides an ideal context for this study due to its active use of limited edition and best-seller marketing strategies. By examining how these cues impact purchase intentions, this research aims to provide insights for marketers to tailor their strategy effectively for consumers in the Netherlands.

The study by Wu and Lee (2016) examined the impact of scarcity and popularity cues on consumer behaviour in online retailing, depending on whether the product is purchased for oneself or as a gift for someone else. They discovered that the scarcity cue "Limited Edition" increases purchase intention more than the popularity cue "bestseller" when consumers buy for themselves. The study also shows that this effect is serially mediated by perceived product uniqueness, which increases the product's perceived value and subsequently increases purchase intention. On the contrary, the popularity cue "Best Seller" increases purchase intention more than this effect is also mediated by perceived consumption risk, which influences the product's perceived value and thus decreases purchase intentions.

Furthermore, the degree of self-other overlap is found to moderate the effect of cue type on purchase intention. Self-other overlap is the closeness of the personal relationship. It moderates so that when the self-other overlap is low, popularity cues lead to an increased purchase intention compared to scarcity cues. These findings are based on the results of several survey-based experimental analyses.

#### 1.2 Problem Statement: Central Research Question and Sub-Questions

While Wu and Lee (2016) provide results on the effect of scarcity and popularity cues on purchase intentions, depending on the consumption target, their study leaves room for further research. Firstly, their study was conducted on adults in the United States almost a decade ago. A lot has changed since then, for example, the influence of the younger Generation Z (Gen-Z), who were not yet adults in 2016. According to Vogue Business, Gen-Z broke the marketing funnel, referring to a very different path to purchase than millennials (Maguire, 2024), and their shopping patterns are different from the generations that came before them. (Noenickx, 2023) The rise of Gen-Z and changed consumer behaviour over time can impact the studied relationships.

Additionally, Wu and Lee (2016) examined relatively low-involvement products that form a low risk for the buyer, namely mugs, bobbleheads, and wine. In contrast, the sneaker market is a different landscape, characterised by its booming growth and higher consumer involvement. Sneakers have become culturally significant as a way of expressing status and wealth. They are now a billion-dollar industry (Grovey & Garza, 2023), where the effects of scarcity and popularity cues on purchase intention, depending on consumption target, might differ. Furthermore, their research was conducted exclusively in an American cultural context, which may not be generalisable to the Netherlands.

Importantly, Wu and Lee (2016) analysed the serial mediation of perceived product uniqueness on the effect of scarcity cues on purchase intention when buying for oneself but did not account for individual differences in the need for uniqueness between consumers. The need for uniqueness is mentioned in their paper, but they only translate this into their research by measuring how the product is perceived to reflect the user's uniqueness. They do not incorporate how much personal need for uniqueness the consumer has. Research by Shin, Eastman, and Mothersbaugh (2017) indicates that a person-specific need for uniqueness significantly affects purchase intentions for limited-edition luxury items. By not considering this, Wu and Lee assume a homogeneous need for uniqueness among consumers.

In contrast, a different need for uniqueness can have a moderating effect on one's purchase intention. Therefore, I will also measure each participant's need for uniqueness and analyse whether there is a moderating effect. Based on these considerations, my thesis will answer the central research question:

How do scarcity (limited edition) and popularity (bestseller) cues influence the purchase intentions of consumers in the Netherlands differently when purchasing sneakers for oneself versus as a gift for others?

I will also answer the following empirical sub-questions:

- 1. Does perceived product uniqueness and perceived product value mediate the effect of scarcity cues on purchase intention for self-purchases?
- 2. Does perceived consumption risk and perceived product value mediate the effect of popularity cues on purchase intention?
- 3. Does the personal need for uniqueness moderate the effect of scarcity cues (vs. popularity cues) on purchase intention for self-purchases?
- 4. Does the degree of self-other overlap moderate the effect of cue type on purchase intention for other-purchases?

To answer the questions mentioned above adequately with my research, I will first collect answers to the following theoretical sub-questions:

- 1. How does purchase intention serve as a measure of consumers' likelihood to buy a product?
- 2. What are scarcity-based promotions (limited edition), and how are they used in marketing strategies to influence purchase intention?
- 3. What are popularity-based promotions (bestsellers), and how are they used in marketing strategies to influence purchase intention?
- 4. How would the consumption target influence the effect of scarcity and popularity cues on purchase intention differently, and what is the role of perceived product uniqueness and consumption risk in this?
- 5. What does the "Limited Edition for Me and Bestseller for You effect" of Wu et al. (2016) entail?
- 6. How would the personal need for uniqueness influence the effect of scarcity and popularity cues on purchase intention?
- 7. How would the degree of self-other overlap influence the effect of scarcity and popularity cues on purchase intention?

#### **1.3 Research Relevance**

#### 1.3.1 Academic Relevance

Academically, this study contributes to the existing knowledge on consumer behaviour and marketing strategies by exploring the interaction between scarcity vs. popularity cues and the consumption target. While there are various studies regarding scarcity and limited edition as marketing tools, to my knowledge, there has not been any research done in the fashion industry specifically for sneakers into the comparison between the cue types, scarcity (limited edition) and popularity (bestseller) on purchase intention, depending on the consumption target.

This study builds further on similar research by Wu and Lee, 2016 by first addressing a literature gap by focusing on a high-involvement product category in the fashion industry (sneakers). Secondly, by researching in a culturally different, non-American context (the Netherlands). Thirdly, it incorporates the personal need for uniqueness as a moderating variable, accounting for individual differences in the need for uniqueness instead of assuming a homogenous need for uniqueness. These contributions are valuable for academics and researchers in marketing, psychology, and consumer behaviour.

#### 1.3.2 Managerial Relevance

The managerial relevance of this study lies in its potential to provide insights for marketers that are applicable specifically within the sneaker industry, as well as in the marketing of other fashion items. By understanding how the cue types scarcity (limited edition) and popularity (bestseller) influence purchase intentions in the different contexts of self-purchase compared to other-purchases, marketers can apply different cue types when aware of the context of the purchases. For example, knowing that scarcity cues might drive higher purchase intentions for self-purchases can help managers plan limited-edition releases more effectively. Similarly, understanding the impact of popularity cues on gift purchases can inform strategies for holiday seasons or other special events. This research can thus offer relevant knowledge, especially to online retailers, to better target marketing practices and promotional strategies to individuals, depending on the purchasing context, demographics, and personal need for uniqueness.

#### 1.3.3 Societal Relevance

This study can be relevant for public policy or educational initiatives promoting responsible consumption and conscious financial decision-making by providing insights into how scarcity and popularity cues influence purchase intentions. These insights can help develop programs that educate consumers about marketing tactics and encourage more thoughtful purchasing decisions, helping prevent people who cannot afford certain products from falling for manipulative marketing tactics. In a time where digital platforms are the norm, consumers often encounter "hunger marketing" strategies that create artificial demand through limited availability, leading to impulsive purchases (Cuofano,

2024; How Temu Uses Psychological Hacks To Encourage Overspending, 2015). Educating consumers about these tactics supports critical thinking and resistance to such manipulations, as they are more aware of how these marketing strategies can be used. This can result in more balanced consumption.

#### 1.4 Research Structure

This paper is further organised as follows: Chapter 2 contains the theoretical framework with a literature review, leading to the hypotheses stated at the end of the chapter. Chapter 3 contains the data and methodology, including the sampling method, data analysis techniques, and the software used for this analysis. Chapter 4 presents the results and their interpretation. Chapter 5 contains this study's conclusions, managerial implications, and limitations. Finally, the appendices include supplementary material such as all four survey versions and more extensively presented data and results.

#### 2 Literature Study

#### 2.1 How does purchase intention serve as a measure of consumers' likelihood to buy a product?

Purchase intention is a measure of rating consumers' stated purchase likelihood, defined as the extent to which customers are willing to buy a product or service (Jamieson & Bass, 1989). It has been used extensively as a metric in marketing research to determine how consumers behave before actual purchasing occurs. It is relevant as it informs marketers about how prepared and interested consumers are in products. Understanding the drivers of purchase intention can help businesses adjust their marketing strategies to meet consumer needs and preferences better, potentially increasing sales.

According to Azjen and Fishbein (1977), purchase intention is highly correlated with the actual purchase behaviour of consumers and is a potential indicator that the consumer's intention leads to purchase behaviour. This relationship is grounded in the Theory of Planned Behavior, which states that behavioural intentions are the most immediate determinants of behaviour (Ajzen, 1991). Multiple studies have supported this theory, demonstrating that higher purchase intentions typically lead to higher actual purchase rates. Sheeran (2002) explores the relationship between intentions and actual behaviours by analysing the extent to which intentions predict behaviours. Sheeran identifies various factors that influence this relationship and finds that while a gap exists, purchase intentions significantly predict actual purchase behaviour. Lee and Lee (2015) also analyse the relationship between stated purchase intentions and actual purchase behaviour in e-commerce. Their main finding is that the process of forming purchase intentions and behaviour is influenced by the expected product value and the actual probability of achieving that value. Consumer intentions indicate the degree to which individuals are willing to perform a specific behaviour, which would be interpreted as online purchasing behaviour in this study. (Ajzen, 1991)

Online purchase intention has become increasingly important nowadays, while e-commerce continues to grow. A meta-analysis by Wright and Macrae (2007) showed that purchase intention scales are empirically unbiased with lower variability than assumed before, improving confidence in their use as a predictor of purchase behaviour. Given its strong correlation with actual purchase behaviour, purchase intention is a valuable metric in marketing research. It helps in understanding consumer readiness to buy, forecasting sales, and evaluating the effectiveness of marketing strategies. Thus, in this study, online purchase intention is used, as it can be seen as the antecedent of purchase behaviour, which refers to a consumer's plan or willingness to buy a particular product through an online store. (Pavlou, 2003)

# 2.2 What are scarcity-based promotions (limited edition), and how are they used in marketing strategies to influence purchase intention?

Scarcity-based promotions are marketing strategies that emphasise the limited availability of a product to create a sense of urgency and exclusivity. Limited edition promotions fall under this category, where

companies intentionally restrict the number of items produced or available for sale. This tactic plays on the scarcity principle, which states that opportunities seem more valuable when they appear more limited (Cialdini, 2006). Research by Gierl, Plantsch, & Schweidler (2008) classifies scarcity cues into supply versus demand and quantity versus time. Marketers create scarcity using various techniques (e.g., limited quantity, limited time, limited edition, only available until... and temporarily available) (Gierl et al., 2008). While limited-time appeals are common, limited-quantity appeals, which include limited editions, are more effective in influencing purchase intentions (Aggarwal, Jun, & Huh, 2011). Limited quantity (e.g., only three left) is a demand-side form of scarcity. In contrast, limited edition (e.g., this product is a limited edition) is considered a supply-side type of scarcity (Gierl et al., 2008).

In the context of the sneaker market, limited edition promotions are especially impactful. Companies often release limited editions to satisfy consumers' desire for exclusivity (Balachander and Stock, 2009), especially in the sneaker market. Consumers look for products that offer a sense of individuality in addition to utility (Eom & Lee, 2012). According to Snyder and Fromkin (1977), people like to express their personality or identity through unique items others do not own, a phenomenon known as the 'snob effect.' This drives many sneaker brands to adopt limited-edition marketing strategies. Limited edition sneakers create a perception of exclusivity and significantly enhance brand value (Gierl et al., 2008; Wu & Lee, 2016); the current study focuses on this particular type of scarcity cue.

The relationship between scarcity cues and purchase intention is driven by the perception of exclusivity and urgency that limited-edition products create. The theoretical foundations of scarcitybased promotions can be explained through commodity and signalling theories. Commodity theory suggests that people value items more when they are challenging to obtain or unavailable (Brock, 1968; Lynn, 1991). This theory helps explain why limited editions positively impact consumer responses, such as product desirability, brand evaluation, perceived value, and purchase intention (Aggarwal & Vaidyanathan, 2003; Eisend, 2008; Ku et al., 2012). However, in the sneaker market, signalling theory is more relevant. This theory presumes that luxury limited-edition products signal high quality and value to consumers because brands make credible commitments by producing a limited number of items (Stock & Balachander, 2005). These products signal the consumer's uniqueness, exclusivity, and social status (Eisend, 2008; Gierl & Huettl, 2010), and in this way positively influence purchase intention.

# 2.3 What are popularity-based promotions (bestseller) and how are they used in marketing strategies to influence purchase intention?

Popularity-based promotions are marketing strategies that show the popularity of a particular product in terms of the number of times it has been purchased, using social proof to enhance its attractiveness. This type of promotion frequently uses tags or claims such as "bestseller," "most popular," or "75% of customers who viewed this also bought this.", which have been extensively used by retailers like Amazon.com and Overstock.com (Wu & Lee, 2016)

These cues give information about general market preferences, acting as social norms that drive a consumer's purchasing behaviour (Berger & Heath, 2007; Fromkin & Snyder, 1980; Goldstein, Cialdini, & Griskevicius, 2008). They provide social validation for product quality, meaning it must be good if many people prefer or have purchased the product (Goldstein et al., 2008; Griskevicius et al., 2009).

Aggarwal, Jun, and Huh (2011) analysed how popularity-based promotions affected consumer behaviour and found that these promotions significantly increase purchase intentions by creating a sense of reassurance and lowering perceived risk.

Popularity-based promotions can work well for marketers in the sneaker market. By advertising a sneaker as a "bestseller," brands can capitalise on the social proof phenomenon, making the product more appealing to consumers who are influenced by the buying behaviours of others. Thus, I expect that popularity cues positively influence purchase intention.

# 2.4 How would the consumption target influence the effect of scarcity and popularity cues on purchase intention differently, and what is the role of perceived product uniqueness and consumption risk in this?

As mentioned before, scarcity and popularity cues are common tactics used in retail to influence customers. However, these strategies are incompatible because they are based on conflicting theories of providing uniqueness or the opposite, namely generality. (Deval et al. 2013; Steinhart et al. 2014) Therefore, they are rarely used together (Gierl et al., 2008). Psychological research shows that specific purchasing goals, such as self-expression and individuation, are more relevant to decision-making regarding oneself. (Abele & Wojciszke, 2007) These factors can be less relevant for decision-making regarding purchases for someone else, and other factors may play a role. It is, therefore, interesting to assess how the consumption target influences the effect of scarcity and popularity cues on purchase intentions.

Purchasing something for yourself is entirely different from buying a gift for another person. This is because there are different psychological drivers behind purchasing decisions. Two essential drivers for consumer behaviour are the need for a unique product and consumption risk.

Perceived product uniqueness refers to the consumer's perception of how distinct and unique a product is compared to others. It is a psychological assessment where consumers believe that the product offers something unique that is not commonly available. Consumers assess the uniqueness of a product based on its rarity, design, features, or any other attribute that sets it apart from standard offerings.

The need for uniqueness and belonging are two basic human needs that influence our daily social interactions. (Snyder & Fromkin 1977; Tian, Bearden, & Hunter 2001). According to social psychology studies, self-related decision-making is more closely associated with agentic goals like individuation and self-expansion (Abele & Wojciszke, 2007). Therefore, in the context of self-purchases, the need for uniqueness will likely outperform the need for belonging and will be an essential consideration when buying for oneself.

In the meantime, when making decisions about others, communal goals, including social integration and consideration for others, take priority (Abele & Wojciszke, 2007). Perceived consumption risk refers to the potential adverse outcomes that a consumer anticipates when purchasing a product. It includes uncertainties about product performance and the fear of negative consequences. When purchasing for others, this risk is even more important as there is a larger uncertainty because the consumer does not know if the recipient will well regard the product. Therefore, it can be expected that perceived consumption risk plays a vital role in the consumer's decision-making process for other-purchases.

Both perceived product uniqueness and perceived consumption risk can be expected to influence purchase intention via perceived value. Perceived value is a concept that refers to a consumer's assessment of the benefits and costs of a product. Perceived value is defined as the trade-off between perceived quality and perceived sacrifice (Wu & Lee, 2016; Gan & Wang, 2017). The evaluation of perceived value includes various aspects such as quality, price, and convenience. When consumers perceive a high value in a product, their intention to purchase increases because they believe the benefits outweigh the costs (Zeithaml, 1988). It is important to note that while high prices can sometimes indicate better quality, they can also diminish the perceived value if they surpass consumers' expectations of what they should pay. This demonstrates the complex relationship between pricing, perceived value, and consumer decision-making. (Dodds et al., 1991). This perception of value is important in influencing consumer behaviour, as higher perceived value tends to lead to higher purchase intentions, ultimately impacting actual purchasing decisions.

### 2.4 What does the "Limited Edition for Me and Bestseller for You effect" of Wu et al. (2016) entail?

As follows from the literature mentioned above, the need for uniqueness is expected to play a more prominent role when purchasing a product for oneself. Scarcity-based promotions, compared to popularity-based ones, are more effective in enhancing feelings of uniqueness and distinctiveness (Tian, Bearden, & Hunter 2001). While popularity-based promotions highlight that the product is appealing to many people, this often diminishes the exclusivity of a product (Tian et al., 2001). Thus, for self-purchases, scarcity cues (compared to popularity cues) heighten the perceived uniqueness of the product, leading to an increased perceived value and stronger purchase intentions. (Wu & Lee, 2016)

On the contrary, purchases made for others are related to a higher consumption risk. Research indicates that popularity-based promotions effectively reduce perceived consumption risk by providing social validation. Scarcity-based promotions emphasise uniqueness but can also indicate higher consumption risk (Tian et al., 2001). Therefore, popularity cues (vs. scarcity cues) lower the perceived consumption risk for other-purchases, increasing perceived value and purchase intentions. Thus, like Wu & Lee, 2016 I also expect the *"Limited Edition for Me and Bestseller for You effect"*.

Based on these insights, I propose the following hypotheses:

#### **Self-purchases**

H1. For self-purchases, scarcity (vs. popularity) cues lead to increased purchase intentions.

**H2.** For self-purchases, the serial mediation effect of perceived product uniqueness and perceived product value explains the positive impact of scarcity (vs. popularity) cues on purchase intentions.

#### **Other-purchases**

H3. For other-purchases, popularity (vs. scarcity) cues lead to increased purchase intentions.

**H4.** For other-purchases, the serial mediation effect of perceived consumption risk and perceived product value explains the positive impact of popularity (vs. scarcity) cues on purchase intentions.

# 2.5 How would the personal need for uniqueness influence the effect of scarcity and popularity cues on purchase intention?

The personal need for uniqueness can be defined as the need to distinguish oneself from others through buying, using, and showing off consumer goods (Tian et al., 2001). Research by Shin, Eastman, and Mothersbaugh (2017) shows that a person-specific need for uniqueness significantly affects purchase intentions for limited-edition luxury items. Every consumer is different, so we cannot assume a homogeneous need for uniqueness. Thus, we propose that consumers with a high need for uniqueness are more likely to respond positively to limited edition products, as these items satisfy their desire for exclusivity and distinctiveness. Consequently, the effect of scarcity cues on purchase intention is amplified among individuals with a strong need for uniqueness. This study proposes that the personal need for uniqueness will moderate the relationship between scarcity cues and purchase intentions, enhancing the "Limited Edition for Me" effect.

**H5.** For self-purchases, personal need for uniqueness moderates the effect of cue type on purchase intention such that scarcity cues (vs popularity cues) increase purchase intention only when the need for uniqueness is high.

# 2.6 How would the degree of self-other overlap influence the effect of scarcity and popularity cues on purchase intention?

Everyone is different. We each have different values and preferences. The degree of self-other overlap is likely to influence purchasing decisions. Self-other overlap can be defined as the extent to which individuals perceive themselves as having a close relationship with the other person and having similarities in their identity. (Myers & Hodges, 2011) Consumption risk increases when purchasing for others, especially if you do not know them well or have different identities. The opposite holds that when the purchaser and recipient share similar identities and have a close relationship, then the

consumption risk decreases. When self-other overlap is high, the decision-making process could be completely different from the case discussed previously. It could resemble a purchasing decision for oneself. Therefore, I propose that the "popularity for others" effect could be moderated by self-other overlap. According to Tu, Shaw, and Fishbach (2015), when interpersonal overlap is high, individuals become more aware of the "self-other collective" during decision-making. In close relationships with significant self-other overlap, consumers are likely to consider themselves and the relational other as a single unit. This leads to egocentric projection, where consumers attribute their values and thoughts to the other person (Murray et al., 2002). The "popularity for others" effect works well when there is little overlap between self and others. Nevertheless, when there is much overlap, decision-making for oneself and others can blend, potentially weakening the "popularity for others" effect to weaken when self-other overlap is high.

**H6.** For other-purchases, the degree of self-other overlap moderates the effect of cue type on purchase intentions, such that popularity cues (vs. scarcity cues) lead to increased purchase intentions only when self-other overlap is low.

#### 2.7 Conceptual Model

Figure 1 displays the conceptual model of the hypotheses. The blue arrows display the hypotheses for the consumption target self-purchases and the red arrows for other-purchases.



Figure 1: Conceptual Model

#### **3** Research Methodology

#### 3.1 Data Collection: Qualitative or Quantitative research

There are two primary methods in academic research: quantitative and qualitative research. Subjective insights are the main emphasis of qualitative research, which includes techniques like focus groups, indepth interviews, and qualitative observation to collect exploratory data on attitudes and feelings. On the other hand, quantitative research uses statistical tests to evaluate findings and depends on numerical data to establish conclusions. According to Malhorta and Birks (2007), there are four kinds of quantitative research: descriptive, correlational, causal-comparative, and experimental. Common quantitative methods include surveys, scientific experiments, and numerical data analysis. For this study, quantitative research was chosen to reach conclusions regarding the effects of scarcity and popularity cues on purchase intention. Quantitative methods offer scientific credibility, especially when dealing with measurable data (Lakshman et al., 2000). Quantitative research also enhances the validity of results by involving larger samples, which contributes to more generalisable findings.

This study will be executed via an online survey on the Qualtrics platform using a  $2 \times 2$  betweensubjects experimental design. This online survey allows for quick distribution via a link or QR code and efficient data collection.

#### 3.2 Survey Overview

Participants will be randomly assigned to one of the four experimental conditions and asked to imagine purchasing a sneaker from an online retailer, either a limited edition for themselves, a limited edition for someone else, a best seller for themselves or a best-seller for someone else, namely a friend. The survey will start by asking to determine sneaker purchase frequency and whether the respondent is a sneaker fanatic. If the respondent never buys sneakers, the survey ends for them as they do not belong to the research target sample. This is followed by a question of to what degree the respondent is a sneaker fanatic. After these, some general demographic questions are asked (i.e., age, gender, and education level). Then, the survey distribution is randomised, and the respondents will be assigned to one of the four conditions. The product description will either include the scarcity cue that the item is a limited edition or the popularity cue that the item is a bestseller and will either be for a purchase for oneself (self-purchase) or a friend (other-purchase).

Table 1: Sneaker description for cue types (bestseller vs. limited edition) and consumption target (Selfvs. Other)

Condition	Description
1. Limited edition – Self purchase	Imagine you want to purchase sneakers for yourself and when
	you are browsing online, you read the following description: This

	exclusive sneaker is a limited-edition release. It features a
	unique design and special colour ways that are not available in
	regular models. This <b>limited-edition sneaker</b> embodies
	exclusivity. Owning this pair means you will stand out with a rare
	and highly sought-after item that only few people possess.
2. Bestseller – Self Purchase	Imagine you want to purchase sneakers for yourself and when
	you are browsing online, you read the following description: This
	sneaker is a bestseller, loved by thousands of customers around
	the world. Renowned for its timeless design, comfort, and
	durability, this sneaker has become an iconic staple in the
	sneaker world. This model is celebrated for its versatility,
	making it a favourite of many people. Owning this pair means
	you are part of a large community of satisfied fans who
	appreciate the classic style and its reliable performance.
3. Limited edition – Other	Imagine you want to purchase a pair of sneakers as a <b>gift</b> for one
purchase	of your friends. Clearly envision who you are buying the gift
	for.
	When you are browsing online for sneakers, you read the
	following description: This exclusive sneaker is a <b>limited-edition</b>
	release. It features a <b>unique</b> design and special colourways that
	are not available in regular models. The limited-edition sneaker
	embodies exclusivity. Owning this pair means your friend will
	stand out with a <b>rare</b> and highly sought-after item that only few
	people possess.
4.Bestseller – Other Purchase	Imagine you want to purchase a pair of <b>sneakers</b> as a <b>gift for one</b>
	of your friends. Clearly envision who you are buying the gift
	for.
	When you are browsing online for sneakers, you read the
	following description: This sneaker is a bestseller, loved by
	thousands of customers around the world. Renowned for its
	timeless design, comfort, and durability, this sneaker has become
	an iconic staple in the sneaker world. This model is celebrated
	for its versatility, making it a <b>favourite of many people</b> . Owning
	this pair means your friend will be part of a large community of
	satisfied fans who appreciate the classic style and its reliable
	performance.

All the participants will be asked about their purchase intention, measured by a three-item measure for purchase intention from Steinhart et al. (2014). Perceived value, the trade-off between perceived quality and perceived sacrifice, will be measured using a single-item scale. (Wu & Lee, 2016)

Participants of the two *self-purchase* variations (Conditions 1 and 2) of the survey will be asked about perceived product uniqueness, defined as the extent to which consumers view the product as different from other products, which will be measured using a two-item scale developed by Franke and Schreier (2008). Furthermore, these participants will be asked additional questions to identify their need or desire for uniqueness based on Zhan and He (2012) and Shin et al. (2017).

The participants of the two *other-purchase* survey variations (Condition 3 and 4) will be asked about perceived consumption risk, defined as how risky they consider the product in a way that the receiver might not like it. Lastly, these participants will be asked to rate their self-other overlap with their chosen friend using the Inclusion of Other in the Self (IOS) scale (Aron, Aron, & Smollan, 1992). For all the above survey items, respondents will be asked to indicate their level of agreement towards each statement using a 7-point Likert scale (1=strongly disagree, 7= strongly agree)

#### 3.3 Research Sample

The online survey targeted sneaker consumers in the Netherlands. The survey was distributed via social channels as private messages to the author's social network. The convenience sampling method was used, which involves selecting the candidate pool that is most conveniently available while still ensuring that each candidate meets the requirements. Additionally, snowball sampling occurred as respondents were requested to forward the survey to others they know. This resulted in a more diverse sample than previously expected, which also included respondents from older age groups. The survey was distributed in June and July 2024. In total, 162 respondents completed the survey across the four experimental conditions. The raw data from the survey can be found in Appendix C.

#### 3.4 Data Analysis

This data will be analysed using STATA and SPSS, as they are reliable and would efficiently perform all the necessary tasks. The data was first reorganised in Stata and split into the correct conditions (1-4). Firstly, the descriptive statistics were calculated using Stata. The number of observations, mean, and standard deviation were calculated for the relevant variables. Secondly, independent t-tests are performed to test differences in purchase intentions between the different cue types (scarcity vs. popularity) and consumption targets (self vs. other). This was done separately for self-purchases (H1) and other-purchases (H3) to determine whether the differences in means between groups were statistically significant. Afterwards, the organised data separated for self and other-purchases was imported and analysed in SPSS. The PROCESS Model 6 (Hayes, 2013) was used for the serial mediation analyses. This tested the serial mediation effects of cue type on purchase intention via perceived product uniqueness and perceived product value (H2) and the serial mediation effect via perceived consumption

risk and perceived product value (H4). For each hypothesis of the mediation analysis, the indirect effects were measured by bootstrapping at 95 percent confidence interval to determine the significance of the mediation paths. The total, direct, and indirect effects are reported to explain the mediation process. Lastly, a moderation analysis using the PROCESS Model 1 (Hayes, 2013) was used to examine whether the need for uniqueness (H5) and self-other overlap (H6) moderates the effects of cue type on purchase intentions. The coefficients and p-values are finally interpreted to assess whether the mediation analysis are significant and aligned with the proposed hypotheses.

#### 4 Research Outcome

#### 4.1 Survey Reliability

The survey was created using the Qualtrics program and was completed by 162 respondents. After cleaning the data, 139 entries were included in the final analysis. The target population for this survey consists of individuals who purchase sneakers at least once a year, as the study aims to understand how consumers behave towards different types of marketing cues and differing consumption targets. To guarantee that the sample adequately represented the target audience, the five respondents who stated they had never purchased sneakers were removed and were not allowed to complete the survey. Additionally, six respondents who completed the survey in under 90 seconds and 12 who took over 600 seconds were excluded. This decision was made to improve the data quality, as very short completion times could indicate they needed to pay more attention and rushed through the responses. On the other hand, very long times could suggest interruptions or a lack of engagement, both of which could compromise the reliability of the data. The distribution of respondents across the four different versions of the experiment after data cleansing was well-balanced due to the randomisation process. Specifically, Condition 1 included 23,74% of respondents, Condition 2 included 25,90%, Condition 3 included 25,18%, and Condition 4 included 25,18%. (Table 2) Because of this balanced distribution, the comparison between different conditions is fair and unbiased, contributing to the robustness of the study's findings. Multiple responses from the same IP addresses were kept, as the survey was sent on various occasions to people using shared internet connections at workplaces and universities. Based on trust, it was assumed that each participant completed the survey only once.

#### 4.2 Survey Outcome Demographics

The distribution between genders shows a slightly higher proportion of females (53.33%) than males (46.67%). The age distribution indicates that the largest group of respondents (28.06%) falls within the 28-34 age range. Notably, the 18-22 and 22-27 age groups collectively represent 35.97% of the sample, who belong to Generation Z. The first three age groups, which are 18-22, 22-27 and 28-34, together, account for over 64,03% of the total sample. The oldest four groups together are 35,98 %. This distribution is skewed compared to the actual population in the Netherlands, with a higher share of younger individuals present. (Table 3)

Table 3 Age Group Distribution

Age Group	Percentage
18-22	25.90%
22-27	10.07%
28-34	28.06%

35-44	14.39%
45-54	5.04%
55-64	0.72%
65 or older	15.83 %

Regarding the distribution of education levels, Table 4 shows that the largest group of respondents holds a bachelor's degree (40.29%), followed by those with some college education (21.58%) and master's degrees (19.42%). High school graduates comprise 17.27% of the sample, while those with professional or doctorate degrees constitute 1.44%. This survey was distributed via convenience sampling, resulting in a higher share of respondents with an academic background.

Table 4 Education Group Distribution
--------------------------------------

Age Group	Percentage
High school graduate	17.27%
Some college (1–4 years, no degree)	21.58%
Bachelor's degree	40.29%
Master's degree	19.42%
Professional or doctorate degree	1.44%

#### 4.3 **Hypothesis** 1

H1. For self-purchases, scarcity (vs. popularity) cues lead to increased purchase intentions.

The first independent t-test only considers self-purchases, with purchase intention as the dependent variable and cue type as the independent variable, where cue type = 0 for the popularity cue and cue type = 1 for the scarcity cue. The sample of popularity cue for self-purchases has 36 observations with  $M_{popularity} = 3.33$  and SD = 1,39. Conversely, the sample of scarcity cues for self-purchases has 33 observations with  $M_{\text{scarcity}} = 4.07$  and SD = 1.58. (Table 5 & 6) This clearly shows a difference in the means of purchase intention, with scarcity cues being more effective than popularity cues (4.07 > 3.33)for self-purchases in influencing purchase intentions. The independent t-test results are  $t_{(1,67)} = -2,06$ with  $p = .04^{**}$ , showing that for self-purchases, there is a significant difference between the means of purchase intention for the different cue types at a 5% significance level. (Table 9) Thus, in line with hypothesis 1, for self-purchase, scarcity (vs. popularity) cues lead to increased purchase intention and hypothesis 1 is accepted. This is visually presented in the first two columns of Figure 1.



Figure 1. Interaction of cue type and consumption target

#### 4.4 Hypothesis 2

H2: For self-purchases, the serial mediation effect of perceived product uniqueness and perceived product value explains the positive impact of scarcity (vs. popularity) cues on purchase intentions.

The Hayes model 6 serial mediation analysis only considers self-purchases, with purchase intention as the dependent variable and cue type as the independent variable, where cue type=0 for the popularity cue and cue type=1 for the scarcity cue. The first tested mediator is perceived product uniqueness, and the second mediator is perceived product value. The sample contains 69 observations for self-purchases. Figure 2 displays the results of all *direct effects*, which are discussed first.

#### 4.4.1 Direct effects

Firstly, when considering the first mediator, *perceived product uniqueness* as the outcome variable, the effect of cue type on perceived product uniqueness is B=1.89 with p-value= $.00^{***}$ , showing scarcity cues (vs. popularity cues) positively affect perceived product uniqueness at a 1% significance level. Scarcity cues are, therefore, associated with higher perceived product uniqueness compared to popularity cues. (Figure 2, a)

Secondly, when considering the second mediator, perceived product value, as the outcome variable, the effect of perceived product uniqueness on perceived product value is B = 0.59 with a p-value =  $.00^{***}$ . (Figure 2, b) Conversely, the effect of cue type on perceived product value is B = 0.18 with a p-value = .66. (Figure 2, c) Thus, perceived product uniqueness positively affects perceived product value at a 1% significance level, but cue type does not significantly affect perceived product value.

Thirdly, when considering purchase intention as the outcome variable, the effect of perceived

product uniqueness on purchase intention is B=0.45 with p-value =.00\*\*\*. (Figure 2, d) Thus, perceived product uniqueness positively affects purchase intention at a 1% significance level. However, the effect of perceived product value on purchase intention is B=0.10 with p-value =.42. (Figure 2, f) This means perceived product value does not significantly influence purchase intentions.



Figure 2: Direct of effects of Hayes Model 6 for Hypothesis 2

#### 4.4.2 Total and Direct effects

The *total and indirect effects* subsequently discussed are displayed in Tables 11 and 13. The total effect of cue type on purchase intention, meaning the effect without considering any mediators, is significant at a 5% level and has an effect of 0.74 with p-value =  $.04^{**}$ . To assess the significance of the mediators, this should be compared to the direct effect of cue type on purchase intention. (Figure 2, e) The direct effect, meaning the remaining effect of cue type on purchase intention after including both mediators, is B= -0.24 with p-value = .55. Thus, this remaining effect is no longer significant. As the relationship between cue type and purchase intention is significant at first, displayed by the significant *total effect* at a 5% significance level, but no longer significant after including the mediators, visible by the insignificant *direct effect* of cue type on purchase intention. It can be concluded that some of the included mediators at least mediate the effect of cue type on purchase intention.

#### 4.4.3 Indirect effects bootstrapped at 95 percent

Finally, looking at the *indirect effects* bootstrapped at 95% for results at a 5% significance level, the first indirect effect of cue type on purchase intention, mediated only by perceived product uniqueness on purchase intention, is significant. (B= 0.85, bootstrapped 95 percent CI: 0.27, 1.52.) This shows that perceived product uniqueness alone already mediates the relationship between cue type and purchase intention. However, the second indirect effect (cue type on purchase intention, mediated by only perceived value) is insignificant. (B: 0.02, bootstrapped 95 percent CI: -0.09, 0.27) Similarly, the third indirect effect, the serial mediation effect of hypothesis 2 (the effect of cue type on purchase intention, mediated by both perceived uniqueness and perceived value), is not significant. (B: 0.11, bootstrapped

95 percent CI: -0.16, 0.45) This shows that perceived product value, alone or as part of a serial mediation, does not significantly mediate the relationship between cue type and purchase intentions. (Tables 13 and 14)

Indirect	Effect	BootSE	BootLLCI	BootULCI
Effect				
Ind1	0.85	0.32	0.27	1.52
Ind2	0.02	0.09	-0.09	0.27
Ind3	0.11	0.15	-0.16	0.45

Table 13 Indirect Effects of Cue Type on Purchase Intention 95% Bootstrap

Table 14 Indirect Effect Paths

Indirect	Path
Effect	
1	Cue Type -> Perceived Product Uniqueness -> Purchase Intention
2	Cue Type -> Perceived Product Value -> Purchase Intention
3	Cue Type -> Perceived Product Uniqueness -> Perceived Product Value -> Purchase Intention

In conclusion, the results show that while perceived product uniqueness significantly mediates the relationship between cue type and purchase intentions, the hypothesised serial mediation through perceived product uniqueness and product value is insignificant. Therefore, Hypothesis 2 is rejected. However, it is important to note that product uniqueness itself is found to mediate the relationship of cue type on purchase intention for self-purchases. This means that scarcity cues (limited edition) compared to popularity cues (bestseller) increase purchase intentions for self-purchases because of a higher perceived product uniqueness.

#### 4.5 Hypothesis 3

H3. For other-purchases, popularity (vs. scarcity) cues lead to increased purchase intentions.

The first independent t-test only considers other-purchases, with purchase intention as the dependent variable and cue type as the independent variable, where cue type = 0 for the popularity scarcity cue and cue type = 1 for the scarcity cue. The sample of popularity cue for other-purchases has 35 observations with  $M_{popularity}$ = 4.72 and SD= 1.56. Conversely, the sample of scarcity cues for other-purchases has 35 observations with  $M_{scarcity}$ = 4.02 and SD= 1.73. (Table 5-6) This clearly shows a difference in the means

of purchase intention, with popularity cues being more effective than scarcity cues (4.72>4.02) for otherpurchases. The results are  $t_{(1,68)} = 1.79$ ,  $p=.08^*$ , showing that for self-purchases, there is a marginally significant difference between the means of purchase intention for the different cue types at a 10% significance level. (Table 10) Thus, in line with hypothesis 3, for other-purchases, there is moderate evidence that scarcity (vs. popularity) cues lead to increased purchase intention, and hypothesis 3 is accepted at a 10% significance level. This is visually presented in the second two columns on the right side of Figure 1.

#### 4.6 Hypothesis 4

H4. For other-purchases, the serial mediation effect of perceived consumption risk and perceived product value explains the positive impact of popularity (vs. scarcity) cues on purchase intentions.

The Hayes model 6 serial mediation analysis only considers other-purchases, with purchase intention as the dependent variable and cue type as the independent variable, where cue type = 0 for the popularity cue and cue type = 1 for the scarcity cue. The first tested mediator is perceived consumption risk, and the second mediator is perceived product value. The sample contains 70 observations for other-purchases. Figure 3 displays the results of all direct effects, which are discussed first.

#### 4.6.1 Direct effects

Firstly, when considering the first mediator, perceived consumption risk as the outcome variable, the effect of cue type on perceived consumption risk is B = 0.86 with p-value= .06, showing scarcity cues (vs. popularity cues) have a marginally significant positive effect on perceived consumption risk at a 10% significance level. Scarcity cues are, therefore, associated with higher perceived consumption risk compared to popularity cues. (Figure 3, a)

Secondly, when considering the second mediator, perceived product value as the outcome variable, the effect of perceived consumption risk on perceived product value is B=-0.25 with p-value =.008\*\*\*. (Figure 3, b) Thus, perceived consumption risk significantly influences perceived product value negatively at a 1% significance level. Additionally, the effect of cue type on perceived product value is B=0.44 with p-value =.20 (Figure 3, c). This means cue type does not significantly influence perceived product value when perceived consumption risk is included in the model.

Thirdly, when considering purchase intention as the outcome variable, the effect of perceived consumption risk on purchase intention is B = -0.38 with p-value =  $.00^{***}$ . (Figure 3, d) The effect of perceived product value on purchase intention is B=0.34 with p-value =  $.007^{***}$ . (Figure 3, f) Thus, perceived consumption risk and perceived product value significantly affect purchase intention at a 1% significance level.



Figure 3: Direct effects of Hayes Model 6 for Hypothesis 4

#### 4.6.2 Total and direct effects comparison

The *total* and *indirect effects* subsequently discussed are displayed in Tables 15 and 17. The *total effect* of cue type on purchase intention, meaning the effect without considering any mediators, is significant at a 10% level and has an effect of B= 0.70 with a p-value= .08\*. To assess the significance of the mediators, the total effect of cue type on purchase intention should be compared to the direct effect of cue type on purchase intention. The direct effect, meaning the remaining effect of cue type on purchase intention after including both mediators, is found to be B= -0.46 with p-value= .18; thus, this relationship is no longer significant after including the mediators. (Figure 3, e) The relationship between cue type and purchase intention is significant after including the mediators, visible by the insignificant *direct effect* of cue type on purchase intention. It can be concluded that at least some of the included mediators significantly mediate the effect of cue type on purchase intention.

#### 4.6.3 Indirect effects bootstrapped at 95 percent

Finally, looking at the indirect effects bootstrapped at 95% for results at a 5% significance level, none of the results are significant. The first indirect effect of cue type mediated only by perceived consumption risk on purchase intention is insignificant. (B= -0.32, bootstrapped 95 percent CI: -0.81, 0.00.) The second indirect effect (cue type on purchase intention, mediated by only perceived value) is also insignificant. (B= 0.15, bootstrapped 95 percent CI: -0.8, 0.42) Similarly, the third indirect effect, the serial mediation effect of hypothesis 4 (the effect of cue type on purchase intention, mediated by both perceived uniqueness and perceived value), is also insignificant. (B= -0.07, bootstrapped 95 percent CI: -0.18, 0.00) (Tables 17 and 18).

Indirect	Effect	BootSE	BootLLCI	BootULCI
Effect				
TOTAL	-0.25	0.27	-0.82	0.24
1	-0.32	0.21	-0.81	0.00
2	0.15	0.13	-0.08	0.42
3	-0.07	0.05	-0.18	0.00

Table 17 Indirect Effects of Cue Type on Purchase Intention 95% Bootstrap

Table 18 Indirect Effect Paths

Indirect	Path
Effect	
1	Cue Type -> Consumption Risk -> Purchase Intention
2	Cue Type -> Perceived Product Value -> Purchase Intention
3	Cue Type -> Consumption Risk -> Perceived Product Value -> Purchase Intention

#### 4.6.4 Indirect effects bootstrapped at 90 percent

Table 19 shows these same indirect effects but bootstrapped at 90% for results at a 10% significance level instead of 5%. The first indirect effect of cue type mediated only by perceived consumption risk on purchase intention is now significant at 10%. (B= -0.32, bootstrapped 95 percent CI: -0.71, -0.03.) The second indirect effect (cue type on purchase intention, mediated by only perceived value) is still insignificant. (B= 0.15, bootstrapped 95 percent CI: -0.04, 0.37) Lastly, the third indirect effect, the serial mediation effect of hypothesis 4 (the effect of cue type on purchase intention, mediated by both perceived consumption risk and perceived value), is significant at the 10% level. (B= -0.07, bootstrapped 90 percent CI: -0.16, -0.004). This shows that perceived consumption risk significantly mediates the relationship between cue type and purchase intention at the 10% level. However, perceived product value does not significantly mediate the relationship between cue type and purchase intentions on its own.

In conclusion, the results show that the hypothesised serial mediation through perceived consumption risk and perceived product value is accepted at a 10% significance level. The coefficient is negative because cue type = 1 for scarcity cues and cue type = 0 for popularity cues, indicating the effect of scarcity (vs. popularity) cues. As hypothesis 4 is about other-purchases, the opposite order is considered, namely how popularity (vs. scarcity) cues influence purchase intention. Thus, the result should be interpreted in the opposite sign for popularity (vs. scarcity) cues, meaning popularity (vs. scarcity) cues positively affect purchase intention. Therefore, Hypothesis 4 is accepted.

Indirect	Effect	BootSE	BootLLCI	BootULCI
Effect				
TOTAL	-0.25	0.26	-0.71	0.16
1	-0.32	0.21	-0.71	-0.03
2	0.15	0.13	-0.04	0.37
3	-0.07	0.05	-0.16	-0.004

Table 19 Indirect Effects of Cue Type on Purchase Intention 90% Bootstrap

#### 4.7 Hypothesis 5

H5. For self-purchases, the degree of personal need for uniqueness moderates the effect of cue type on purchase intention such that scarcity cues (vs popularity cues) increase purchase intention only when the need for uniqueness is high.

The Hayes PROCESS model 1 (Hayes, 2013) moderation analysis only considers self-purchases, with purchase intention as the dependent variable and cue type as the independent variable, where cue type = 0 for the popularity cue and cue type = 1 for the scarcity cue. The moderator is personal need for uniqueness and the sample contains 69 observations for self-purchases. The interaction effect of cue type and need for uniqueness is B = 0.68 with a p-value  $=.002^{***}$ . This shows that the need for uniqueness moderates the effect of cue type on purchase intention. (Table 20)

intention							
Variable		Coefficient	Standard	t-value	p-value	Lower 95%	Upper 95%
			Error (SE)			CI (LLCI)	CI (ULCI)
Constant		3.71	0.17	22.08	0.00	3.37	4.05
Cue Type		0.75	0.34	2.22	0.03	0.07	1.42
Need	for	0.03	0.10	0.34	0.73	-0.17	0.24
Uniqueness	5						
Interaction		0.68	0.20	3.30	0.00	0.27	1.09

Table 20 *Effect of personal need for uniqueness on the relationship between cue type and purchase intention* 

To assess how the level of personal need for uniqueness influences the effect of cue type on purchase intention, the result of a low level (1 SD below the mean), average level (mean) and high level (1 SD above the mean) are compared in table 21. The conditional effects of cue type on purchase intention, meaning the effect when keeping the personal need for uniqueness fixed at a specific mean level, give the following results. At a low level of personal need for uniqueness (1 SD below the mean), the effect

of cue type is B=-0.37 with a p-value=.44, thus negative but insignificant. The conditional effects of cue type at the mean and high levels (1 SD above the mean) are B = 0.75, with a p-value=.03\*\*, thus significant at a 5% significance level and B= 1.86 with p-value=.00\*\*\*, thus significant at a 1% significance level. Based on these results and in line with hypothesis 5, it can be concluded that having an average and higher personal need for uniqueness significantly increases the effect of the scarcity cues (vs. popularity cues) on purchase intention for self-purchases. Thus, hypothesis 5 is accepted.

Need f	for	Effect	Standard		t-value	p-value	Lower	Upper
uniqueness			Error (SE)	)			95% CI	95% CI
							(LLCI)	(ULCI)
-1.65 (M - 1 SD	)	-0.37	0.48		-0.78	0.44	-1.32	0.58
0.00 (M)		0.75	0.34		2.22	0.03	0.07	1.42
1.65 (M + 1 SD)	)	1.86	0.48		3.90	0.00	0.91	2.82

Table 21 *Effect of need for uniqueness on the relationship between cue type and purchase intention for different levels of need for uniqueness* 

#### 4.8 Hypothesis 6

H6. For other-purchases, the degree of self-other overlap moderates the effect of cue type on purchase intentions, such that popularity cues (vs. scarcity cues) lead to increased purchase intentions only when self-other overlap is low.

The Hayes PROCESS model 1 (Hayes, 2013) moderation analysis only considers other-purchases, with purchase intention as the dependent variable and cue type as the independent variable, where cue type= 0 for the popularity cue and cue type= 1 for the scarcity cue. The moderator is self-other overlap, and the sample contains 70 observations for other-purchases. Note that for other-purchases, we found that popularity cues (vs. scarcity cues) significantly increase purchase intention (hypothesis 3). In the model, cue type = 0 for popularity cues and cue type = 1 for scarcity cues, indicating the effect of scarcity (vs. popularity) cues. As hypothesis 6 is about other-purchases, the opposite order is considered, namely how popularity (vs. scarcity) cues influence purchase intention. Thus, the result should be interpreted in the opposite: a positive coefficient means that the corresponding level of self-other overlap decreases the effect of popularity cues (vs. scarcity cues), and a negative coefficient means the level of self-other overlap increases the effect of popularity cues (vs. scarcity cues).

The interaction effect of cue type and self-other overlap is B = 0.56 with p-value = .04\*\*, which is significant at a 5% significance level. Thus, the degree of self-other overlap moderates the relationship between cue type and purchase intention. (Table 22)

Variable	Coefficient	Standard	t-value	p-value	Lower	Upper
	(coeff)	Error (SE)			95% CI	95% CI
					(LLCI)	(ULCI)
Constant	4.32	0.19	22.36	0.00	3.94	4.71
CueType	-0.76	0.39	-1.96	0.05	-1.53	0.02
SOO	0.15	0.13	1.16	0.25	-0.11	0.41
Interaction	0.56	0.26	2.15	0.04	0.04	1.07

Table 22 Effect of self-other overlap on the relationship between cue type and purchase intention

To assess how the level of self-other overlap influences the effect of cue type on purchase intention, the results of a low level (1 SD below the mean), average level (mean) and high level (1 SD above the mean) are compared in table 23. The conditional effects of cue type on purchase intention, meaning the effect when keeping the self-other overlap fixed at a specific mean level, show the following result. At a low level of self-other overlap, namely 1 SD below the mean, the effect of cue type is B= -1.60 with a p-value=.00\*\*\*, thus negative and significant at a 1% significance level. This negative coefficient means that low self-other overlap increases the effect of popularity cues (vs. scarcity cues), which aligns with hypothesis 6. The conditional effects of cue type at the mean and high levels (1 SD above the mean) are B = -0.76 with a p-value=.05 and B=0.09 with a p-value=.87, thus, both are not significant.

These results align with hypothesis 6, and it can be concluded that a low level of self-other overlap significantly increases the effect of popularity (vs. scarcity) cues on purchase intention. Thus, hypothesis 6 is accepted.

Self-other	Effect	Standard	t-value	p-value	Lower	Upper
overlap		Error (SE)			95% CI	95% CI
					(LLCI)	(ULCI)
-1.52 (M - 1 SD)	-1.60	0.56	-2.88	0.01	-2.72	-0.49
0.00 (M)	-0.76	0.39	-1.96	0.05	-1.53	-0.02
1.52 (M + 1 SD)	0.09	0.55	0.17	0.87	-1.00	1.18

Table 23 *Effect of self-other overlap on the relationship between cue type and purchase intention for different levels of self-other overlap* 

### 4.9 Summary of Key Findings

### Table 7 Summary of Key Findings

Hypothesis	Result
H1. For self-purchases, scarcity (vs. popularity) cues lead to increased purchase	Accepted
intentions.	
H2. For self-purchases, the serial mediation effect of perceived product uniqueness	Rejected
and perceived product value explains the positive impact of scarcity (vs. popularity)	
cues on purchase intentions.	
H3. For other-purchases, popularity (vs. scarcity) cues lead to increased purchase	Accepted
intentions.	(10%
	significance)
H4. For other-purchases, the serial mediation effect of perceived consumption risk	Accepted
and perceived product value explains the positive impact of popularity (vs. scarcity)	(10%
cues on purchase intentions.	significance)
H5. For self-purchases, the degree of need for uniqueness moderates the effect of cue	Accepted
type on purchase intention such that scarcity cues (vs popularity cues) increase	
purchase intention only when need for uniqueness is high.	
H6. For other-purchases, the degree of self-other overlap moderates the effect of cue	Accepted
type on purchase intentions, such that popularity cues (vs. scarcity cues) lead to	
increased purchase intentions only when self-other overlap is low.	

#### 5 Conclusion & Recommendations

This chapter begins by outlining the key findings from both the literature study and the results of the empirical research. Following this, the central research question is in the conclusions section. Furthermore, this chapter also discusses any limitations of the research. Finally, it concludes with managerial recommendations and recommendations for future researchers, as well as personal reflections on the thesis journey.

#### 5.1 Key Findings Literature

A lot of research has been done on scarcity cues and consumer behaviour. While there is also some work on popularity cues, it is somewhat less extensive. However, studies comparing these cue types, particularly for different consumption targets, are even more limited.

The literature study found that purchase intention serves as an effective measure of consumers' likelihood to buy a product. Purchase intention is highly correlated with actual purchase behavior and acts as an antecedent to purchasing.

Scarcity-based promotions are marketing strategies that emphasise a product's limited availability to create a sense of urgency and exclusivity. The signalling theory presumes that luxury limited-edition products signal high quality and value to consumers because brands make credible commitments by producing a limited number of items (Stock & Balachander, 2005). These products signal the consumer's uniqueness, exclusivity, and social status (Eisend, 2008; Gierl & Huettl, 2010). Thus, it is expected that scarcity cues positively influence purchase intention.

Popularity-based promotions are marketing strategies that show the popularity of a particular product in terms of the number of times it has been purchased. This type of marketing relies on social proof to enhance product attractiveness. (Goldstein, Cialdini, and Griskevicius, 2008; Griskevicius et al., 2009). By describing a product as a bestseller, these promotions provide social validation and reduce perceived risk, which increases purchase intentions. (Aggarwal, Jun, and Huh 2011) The literature indicates that these popularity cues work effectively because the buying behaviors of others influence consumers.

In the literature review, it is found that the consumption target, whether a purchase is for oneself or a gift for someone else, influences the effect of scarcity and popularity cues on purchase intentions. (Wu & Lee, 2016) Two essential drivers for consumer behaviour are perceived product uniqueness and consumption risk. For self-purchases, the need for uniqueness plays a critical role. Scarcity cues, which are found to enhance perceived product uniqueness, are more effective in increasing purchase intentions for self-purchases. That is why scarcity cues are found to increase purchase intentions more compared to popularity cues for self-purchases. However, for other-purchases, perceived consumption risk becomes more important. Popularity cues reduce this consumption risk for the buyer through social validation, thus increasing purchase intentions for other-purchases. That is why, contrary to selfpurchases, popularity cues are found to increase purchase intentions more compared to scarcity cues for self-purchases. (Wu and Lee, 2016)

The literature review also explains the moderating effects of personal need for uniqueness and self-other overlap. In a research setting different from Wu and Lee (2016), it is found that consumers with a high need for uniqueness respond more to scarcity cues, amplifying their purchase intentions for limited edition products. (Shin, Eastman, and Mothersbaugh, 2017) In the context of other-purchases, the degree of self-other overlap influences the effectiveness of popularity cues. When self-other overlap is low, popularity cues are more effective in increasing purchase intentions due to reduced consumption risk.

Overall, the literature review provides a solid theoretical foundation for understanding how scarcity and popularity cues influence purchase intentions and the underlying processes as to how the consumption target can influence these effects due to perceived consumption risk and product uniqueness. It furthermore explains the importance of individual differences in the need for uniqueness and self-other overlap, all leading to the hypotheses of this study.

#### 5.2 Key Findings Research

The empirical research tested six hypotheses related to the impact of cue type (scarcity vs. popularity) on purchase intentions among sneaker consumers in the Netherlands. These results are based on a sample of 162 respondents who purchase sneakers at least once a year. The participants were randomly assigned to one of four experimental conditions: limited edition for self-purchase, limited edition for other-purchase, bestseller for self-purchase, and bestseller for other-purchase.

The first hypothesis states that scarcity cues (limited edition) increase purchase intention compared to popularity cues (bestseller) for self-purchases. The results showed that the mean of purchase intention for scarcity cues was higher than for popularity cues. The independent t-test confirmed a significant difference between the two means of purchase intention. Thus, in line with hypothesis 1, for self-purchases, scarcity (vs. popularity) cues lead to increased purchase intention and hypothesis 1 is accepted.

The second hypothesis states that the positive impact of scarcity cues on purchase intentions for self-purchases would be mediated by perceived product uniqueness and perceived product value. The results show that while perceived product uniqueness significantly mediates the relationship between cue type and purchase intentions positively, the hypothesised serial mediation through perceived product uniqueness and perceived product uniqueness 2 is rejected.

The third hypothesis states that popularity (bestseller) cues lead to higher purchase intentions for other purchases compared to scarcity cues (limited edition). The results showed that the mean of purchase intention for popularity cues was higher than for scarcity cues. The independent t-test confirmed a marginally significant difference between the two means of purchase intention. Thus, in
line with hypothesis 3, for other-purchases, popularity (vs. scarcity) cues lead to increased purchase intention, and hypothesis 3 is accepted.

The fourth hypothesis states that the positive impact of popularity cues on purchase intentions for other-purchases would be mediated by perceived consumption risk and perceived product value. The results show that the serial mediation effect through perceived consumption risk and perceived product value is marginally significant. Thus, hypothesis 4 is accepted.

The fifth hypothesis states that the need for uniqueness would moderate the effect of cue type on purchase intentions for self-purchases, with scarcity cues increasing purchase intentions more compared to popularity cues, only when the need for uniqueness is high. The moderation analysis showed a significant interaction effect, confirming that the need for uniqueness moderates the main effect. Based on the results for specifically different levels of need for uniqueness, it can be concluded that having a low need for uniqueness does not have a significant effect. However, in line with hypothesis 5, an average and higher personal need for uniqueness significantly increases the effect of the scarcity cues (vs. popularity cues) on self-purchase purchase intention. Thus, hypothesis 5 is accepted.

Finally, the sixth hypothesis states that self-other overlap moderates the effect of cue type on purchase intentions for other-purchases, with popularity cues as compared to scarcity cues, increasing purchase intentions only when self-other overlap is low. The moderation analysis showed a significant interaction effect, confirming that self-other overlap moderates the effect of cue type on purchase intention. Based on the results for specifically different levels of self-other overlap, it can be concluded that having average or high self-other overlap does not have a significant effect. However, in line with hypothesis 6, a low self-other overlap significantly increases the effect of the popularity cues (vs. scarcity cues) on purchase intention for other-purchases. Thus, hypothesis 6 is accepted.

In summary, the empirical research confirmed the majority of the hypotheses, demonstrating that the consumption target is an important influencer of the effects of scarcity and popularity cues on purchase intentions due to the differential psychological processes of self-purchases and other-purchases. The effects of cue type on purchase intention are mediated by perceived product uniqueness for self-purchases and mediated by consumption risk and perceived product value for other-purchases. Furthermore, the effects of cue type on purchase intention are moderated by individual differences in personal need for uniqueness for self-purchases and by degree of self-other overlap for other purchases.

#### 5.3 Conclusion

This study investigates the effects of scarcity and popularity cues on consumers' purchase intentions depending on the consumption target in the Netherlands, particularly in the context of sneakers. It focuses on the mediators perceived product uniqueness, perceived consumption risk, and perceived value, as well as the moderator's personal need for uniqueness and self-other overlap.

The central research question, as stated in Chapter 1, is as follows:

How do scarcity (limited edition) and popularity (bestseller) cues influence the purchase intentions of consumers in the Netherlands differently when purchasing sneakers for oneself versus as a gift for others?

To answer this research question, an online survey was conducted with a sample of 162 sneaker consumers in the Netherlands. The survey used a  $2 \times 2$  between-subjects experimental design, assigning participants to one of four conditions: limited edition for self-purchase, limited edition for other-purchase, bestseller for self-purchase, and bestseller for other-purchase. Various statistical analyses, including independent t-tests, mediation, and moderation analyses, were performed using Stata and SPSS software.

In the experimental research of this thesis, a clear answer to the main research question is found. Depending on the consumption target, namely, if a consumer is purchasing sneakers for oneself or purchasing sneakers as a gift for a friend, the effect of scarcity (limited edition) and popularity (bestseller) cues on purchase intention has a different relative performance. For self-purchases, scarcity (limited edition) cues increase purchase intentions more than popularity (bestseller) cues. Opposite to this result, for other-purchases, popularity (bestseller) cues increase purchase intentions more than popularity (limited edition) cues. These results are in line with the hypotheses formulated in the theoretical framework. From this finding, it follows that depending on the consumption target of the consumer, a different cue type is the better option in marketing strategies. When consumers are looking to buy for themselves, scarcity cues are proven to be more effective in raising their purchase intention. When buying for someone else, popularity cues are proven more effective. This is a valuable insight for marketers who can choose which cue types to use in their marketing strategies. Especially in the online retail sector, where tailored advertisements can be used, this knowledge of consumer behaviour can be used to the general advantage of marketers and businesses.

Furthermore, this research proceeds to shed light on the reason why scarcity (vs. popularity) cues increase purchase intention more for self-purchases, while popularity (vs. scarcity) cues increase purchase intention more for other-purchases. This is done by analysing mediation effects, which show that for self-purchases, perceived product uniqueness mediates the effect of scarcity (vs. popularity cues) on purchase intention, while for other-purchases, perceived consumption risk and perceived product value mediate the effect of popularity (vs. scarcity) cues on purchase intention. From these results, it can be concluded that in line with the discussed theory, scarcity-based promotions, compared to popularity-based ones, are more successful in increasing feelings of uniqueness and distinctiveness and thus increase purchase intention.

For other-purchases, such scarcity cues indicate a risk that the person receiving the gift might not like it because of a deviation from what is common. Popularity cues, on the other hand, indicate that the product is liked by others and provides social validation. This reduces perceived consumption risk by providing social validation, lowering the consumption risk, and making the product more attractive to purchase.

Finally, this research also analyses the moderation effects of personal need for uniqueness and self-other overlap. Results of the first moderation analysis show that for self-purchases, the personal need for uniqueness moderates the effect of cue type on purchase intention, such that scarcity cues (vs popularity cues) increase purchase intention only when the need for uniqueness is average or high. This shows that the need for uniqueness is not homogenous among consumers and that the differences in personal need for uniqueness influence consumers' sensitivity to scarcity cues compared to popularity cues.

Results of the second moderation analysis show that for other-purchases, the degree of selfother overlap moderates the effect of cue type on purchase intentions, such that popularity cues (vs. scarcity cues) lead to increased purchase intentions only when self-other overlap is low. This shows that when consumers do not consider their relationship with the friend in mind for who they are buying a gift as close and have similar identities, the relative effect of popularity (vs. scarcity cues) on purchase intention diminishes. These additional moderation effects suggest that individual differences significantly impact how consumers respond to marketing cues.

#### 5.4 Research Limitations

This study's use of convenience sampling and informal distribution channels has limitations on the generalisability of the findings to a larger population. Firstly, the sample consists of 162 respondents, which is not enough for a representative sample of the population in the Netherlands. The age distribution of the sample is also skewed, with a higher share of younger respondents than the population in the Netherlands. Additionally, the sample has a higher share of respondents with either a bachelor's, master's or doctorate degree (61,15%), which does not reflect the national average of 33,40% of the Netherlands. (Statista, 2024) Secondly, the conclusions of this study in the sneaker market may be generalisable to other markets, but it cannot be known to what extent it is. Therefore, the unknown validity of these results in other markets is also a research limitation.

Additionally, the context of this study is online retail, not a physical store, so the findings might differ for purchase behaviour in physical stores. Thirdly, the study relied on product descriptions instead of actual products, which might not fully capture consumer behaviour in real life. Lastly responses may have been influenced by assumptions about the product's cost, which could have affected the perceived value and purchase intentions, especially for limited edition items, of which respondents could have thought it too expensive.

#### 5.5 Recommendation for Future Research

For future research, I recommended that the limitations of this study are addressed to find more robust effects of scarcity and popularity cues on consumer behaviour. Increasing the sample size and increasing its diversity in age groups and education levels is necessary to improve the generalizability of the findings. Additionally, a field experiment instead of an online survey is recommended, as this provides more realistic insights into consumer behaviour, because participants then interact with actual products rather than descriptions. The research could also be extended to include other fashion items instead of sneakers, which will broaden the understanding of how these marketing strategies influence purchase intentions in different product categories. If similar results are found, this increases the generalizability of these findings. Furthermore, controlling for price-levels in future studies would help improve the robustness of the effects of scarcity and popularity cues on purchase intentions. This would need a larger split in different additional conditions of the survey, which was not realistic for this study due to time and sample constraints. Finally, replicating this study in other countries would offer a broader perspective. This would enhance the robustness of the findings and for global marketing strategies.

#### 5.6 Managerial Recommendation

The results suggest that managers and marketers in the B2C sectors can increase the effectiveness of their strategies by tailoring marketing messages to emphasise either scarcity or popularity, depending on the potential consumer's consumption target. The following concrete strategies are specifically beneficial.

Firstly, to optimise marketing strategies, online businesses should smartly use gathered online customer information and create profiles based on consumption targets and the extent of consumers' need for uniqueness. Besides regular data collection methods used by online companies on their websites and from larger databases, companies can expand this data collection with small surveys and monitoring of purchase history to determine customers' personal need for uniqueness. This approach based enables businesses to tailor their marketing messages effectively.

Secondly, when, based on the consumer profile, consumers intend to purchase products for themselves, I advise marketers to specifically highlight the product's exclusivity in promotional campaigns and emphasise that the product is a limited edition in the product description on the webpage. Emphasising that the product is a limited edition can significantly boost purchase intentions among these consumers. This is especially the case for consumers with a high need for uniqueness, not for people who do not want to stand out. So, when data on purchasing history suggests that the potential consumer has a high need for uniqueness profile, this strategy is especially recommended.

Thirdly, when consumers intend to purchase products for others, I advise marketers to promote products as bestsellers, as this can significantly increase consumers' purchase intentions when buying gifts for others. Convincing consumers of the product's likeability and widespread acceptance reassures gift buyers of their choice. Based on the results of this research, it is advisable, for example, that during the holiday season or when selling products mostly bought by consumers as a gift, the product's popularity should be promoted more than exclusivity. This is especially the case for gifts purchased for people with whom consumers are less close and different in their identity, which is the case when self-other overlap is low.

Lastly, I advise the management of businesses that sell products mainly bought for selfpurchases to manufacture or offer limited-edition products in their product offers. Many companies never launch limited edition products, while this can boost sales. This research shows scarcity cues are more effective for self-purchases than popularity cues, so in self-purchase dominant market sectors, it is suboptimal only to offer a wide range of products and promote these as being great and loved by many people (best seller cue), as limited-edition products are proven to enhance purchase intention more. Managers should, therefore, critically assess if they are optimally using the potential of limited-edition products in their product launches, and if they do not already do so, make sure to launch a limitededition product occasionally.

In conclusion, businesses should leverage consumer data to develop personalised marketing campaigns. By understanding individual differences, such as the need for uniqueness and self-other overlap, I advise marketers to use different messages in line with triggering the different psychological processes behind scarcity and popularity cues described above.

#### 5.7 Reflection

Writing this thesis has been a valuable learning experience. I have improved my ability to find and quickly evaluate academic articles. My writing skills have also improved, especially in structuring and explaining ideas and relationships, all the way from the literature review through the survey building and interpretation of results to the conclusion. This was especially challenging given the 2x2 research design of the study. I have also developed my skills of analysing data using both the statistical software's Stata and SPSS. This writing process has increased my appreciation for academic research and prepared me for further independent research in my master's studies. The thesis journey has also taught me resilience and the importance of working hard in research. It has finally strengthened my ability to think critically and to solve problems.

## Appendix

### **Appendix A- References**

- A cult classic: Culture and its influence on consumer behaviour. file:///C:/Users/535549tl/Downloads/A Cult Classic Sneaker Cultur.pdf
- Aggarwal, P., Jun, S. Y., & Huh, J. H. (2011a). Scarcity messages. *Journal of Advertising*, 40(3), 19-30.
- Aggarwal, P., Jun, S. Y., & Huh, J. H. (2011b). Scarcity messages. *Journal of Advertising*, 40(3), 19-30.

Ajzen, I. (1991). The theory of planned behavior. *Organizational behavior and human decision* processes, 50(2), 179-211.

- Ajzen, I., & Fishbein, M. (1977). Attitude-behavior relations: A theoretical analysis and review of empirical research. *Psychological Bulletin*, 84(5), 888.
- Balachander, S., & Stock, A. (2009). Limited edition products: When and when not to offer them. *Marketing Science*, 28(2), 336-355.
- Berger, J., & Heath, C. (2007). Where consumers diverge from others: Identity signaling and product domains. *Journal of Consumer Research*, 34(2), 121-134.
- Bozkurt, S., & Gligor, D. (2019). Scarcity (versus popularity) cues for rejected customers: The impact of social exclusion on cue types through need for uniqueness. *Journal of Business Research*, 99, 275-281.
- Brock, T. C. (1968). Implications of commodity theory for value change. *Psychological Foundations of Attitudes*, , 243-275.
- Choi, J. W.Sneakerheads' assessment of sneaker value and behaviors throughout the sneaker ownership cycle. Retrieved Jun 12, 2024, from
- Cialdini, R. B., & Cialdini, R. B. (2007). Influence: The psychology of persuasion. Collins New York.
- Cuofano, W. I. G. (2024, 19 april). *What is hunger marketing? Hunger marketing in a nutshell*. FourWeekMBA. https://fourweekmba.com/hunger-marketing/
- Dodds, W. B., Monroe, K. B., & Grewal, D. (1991). Effects of price, brand, and store information on buyers' product evaluations. *Journal of Marketing Research*, 28(3), 307-319.

- Eisend, M. (2008). Explaining the impact of scarcity appeals in advertising: The mediating role of perceptions of susceptibility. *Journal of Advertising*, *37*(3), 33-40.
- Eom, K. H., & Lee, Y. (2012). Research reports: A study on development of sportswear product design through collaboration. *Journal of the Korean Society of Design Culture, 18*(3), 274-283.
- Gan, C., & Wang, W. (2017). The influence of perceived value on purchase intention in social commerce context. *Internet Research*, 27(4), 772-785.
- Gergen, K. (2012). Social exchange: Advances in theory and research. Springer Science & Business Media.
- Gierl Heribert, & Huettl Verena. Are scarce products always more attractive? the interaction of different types of scarcity signals with products' suitability for conspicuous consumption ScienceDirect. https://www.sciencedirect.com/science/article/pii/S0167811610000285
- Gierl, H., Plantsch, M., & Schweidler, J. (2008). Scarcity effects on sales volume in retail. *The International Review of Retail, Distribution and Consumer Research, 18*(1), 45-61.
- Goldstein, N. J., Cialdini, R. B., & Griskevicius, V. (2008). A room with a viewpoint: Using social norms to motivate environmental conservation in hotels. *Journal of Consumer Research*, 35(3), 472-482.
- Griskevicius, V., Goldstein, N. J., Mortensen, C. R., Sundie, J. M., Cialdini, R. B., & Kenrick, D. T. (2009). Fear and loving in las vegas: Evolution, emotion, and persuasion. *Journal of Marketing Research*, 46(3), 384-395.
- Hayes, A. F. (2017). Introduction to mediation, moderation, and conditional process analysis: A regression-based approach. Guilford publications.
- He, Y., & Oppewal, H. (2018). See how much we've sold already! effects of displaying sales and stock level information on consumers' online product choices. *Journal of Retailing*, 94(1), 45-57.
- Jamieson, L. F., & Bass, F. M. (1989). Adjusting stated intention measures to predict trial purchase of new products: A comparison of models and methods. *Journal of Marketing Research*, 26(3), 336-345.
- Jang, W. E., Ko, Y. J., Morris, J. D., & Chang, Y. (2015). Scarcity message effects on consumption behavior: Limited edition product considerations. *Psychology & Marketing*, 32(10), 989-1001.
- Ku, H., Kuo, C., & Kuo, T. (2012a). The effect of scarcity on the purchase intentions of prevention and promotion motivated consumers. *Psychology & Marketing*, *29*(8), 541-548.

- Ku, H., Kuo, C., & Kuo, T. (2012b). The effect of scarcity on the purchase intentions of prevention and promotion motivated consumers. *Psychology & Marketing*, *29*(8), 541-548.
- Ku, H., Kuo, C., & Kuo, T. (2012c). The effect of scarcity on the purchase intentions of prevention and promotion motivated consumers. *Psychology & Marketing*, *29*(8), 541-548.

Lee, J., & Lee, J. N. (2015). How purchase intention consummates purchase behaviour: the stochastic nature of product valuation in electronic commerce. *Behaviour & Information Technology*, *34*(1), 57-68.

- Lynn, M. (1991). Scarcity effects on value: A quantitative review of the commodity theory literature. *Psychology & Marketing*, 8(1), 43-57.
- Myers, M. W., & Hodges, S. D. (2012). The structure of self-other overlap and its relationship to perspective taking. *Personal Relationships*, 19(4), 663-679.
- Pavlou, P. A. (2003a). Consumer acceptance of electronic commerce: Integrating trust and risk with the technology acceptance model. *International Journal of Electronic Commerce*, *7*(3), 101-134.
- Pavlou, P. A. (2003b). Consumer acceptance of electronic commerce: Integrating trust and risk with the technology acceptance model. *International Journal of Electronic Commerce*, *7*(3), 101-134.

Sheeran, P. (2002). Intention—behavior relations: a conceptual and empirical review. *European review* of social psychology, 12(1), 1-36.

- Snyder, C. R., & Fromkin, H. L. (1977). Abnormality as a positive characteristic: The development and validation of a scale measuring need for uniqueness. *Journal of Abnormal Psychology*, *86*(5), 518.
- Statista. (2024, 5 juli). Population of the Netherlands 2013-2022, by education level. https://www.statista.com/statistics/914489/population-of-the-netherlands-by-educationlevel/#:~:text=Population%20of%20the%20Netherlands%202013%2D2022%2C%20by%20educ ation%20level&text=Between%202013%20and%202022%2C%20the,had%20grown%20to%201 4.2%20percent.
- Steinmetz, J. How temu uses psychological hacks to encourage overspending | psychology today. https://www.psychologytoday.com/intl/blog/the-social-consumer/202404/how-temu-uses-psychological-hacks-to-encourage-overspending
- Wright, M., & MacRae, M. (2007). Bias and variability in purchase intention scales. *Journal of the Academy of Marketing Science*, *35*, 617-624.

Zeithaml, V. A. (1988). Consumer perceptions of price, quality, and value: A means-end model and synthesis of evidence. *Journal of Marketing*, *52*(3), 2-22.

## Appendix B – Tables and Figures

Condition	Frequency	Percent
1	33	23.74%
2	36	25.90%
3	35	25.18%
4	35	25.18%
Total	139	100.00%

Table 2 Distribution of respondents per condition

## Table 3 Age Group Distribution

Age Group	Percentage
18-22	25,90%
22-27	10,07%
28-34	28.06%
35-44	14,39%
45-54	5,04%
55-64	0,72%
65 or older	15,83 %

Table 4 Education Groups Distribution

Age Group	Percentage
High school graduate	17,27%
Some college (1-4 years, no degree)	21,58%
Bachelor's degree	40,29%
Master's degree	19,42%
Professional or doctorate degree	1,44%

Table 5 Condition 1: Consumption Target Self - Limited Edition

Variable	Observations	Mean	Std. Dev.	Min	Max
	(Obs)				
PI_avg	33	4,07	1,58	2,00	7,00
PU_avg	33	4,55	1,44	1,50	7,00
NFU_avg	33	3,98	1,68	1,00	7,00
PV	33	4,97	1,67	1,00	7,00

 Table 6 Condition 2: Consumption Target Self - Bestseller

Variable	<b>Observations</b>	Mean	Std.Dev.	Min	Max
PI_avg	36	3,33	1,39	1,00	5,33
PU_avg	36	2,65	1,32	1,00	5,50
NFU_avg	36	4,13	1,64	1.50	7,00
PV	36	3,67	1,53	1,00	6,00

Table 7 Condition 3: Consumption Target Other - Limited Edition

Variable	Observations	Mean	Std. Dev.	Min	Max
PI_avg	35	4,01	1,73	1,00	6,00
PV	35	5,40	1,47	1,00	7,00
CR	35	4,34	1,73	1,00	7,00
SOO	35	5,40	1,39	2,00	7,00

 Table 8 Condition 4: Consumption Target Other – Bestseller

Variable	Observations	Mean	Std. Dev.	Min	Max
PI_avg	35	4,72	1,56	1,00	7,00
PV	35	5,17	1,38	2,00	7,00
CR	35	3,49	1,95	1,00	7,00
SOO	35	5,06	1,64	1,00	7,00

Table 9 Independent t-test for self-purchases

Group:	Obs	Mean	Std. Err.	Std.	[95% Conf. Interval]
Self-Purchase				Dev.	
Bestseller = 0	36	33.333	0.2323	13.939	2.8617 - 3.8049
Limited Edition =1	33	40.707	0.2751	15.806	3.5102 - 4.6312
Combined	69	36.860	0.1831	15.210	3.3206 - 4.0514
Difference		-0.7374	0.3581		-1.45220.0226
Degrees of Freedom					67
t-test					t = -2.0590, p = 0.0434**

Group:	Obs	Mean	Std. Err.	Std. Dev.	[95% Conf. Interval]
Other-Purchase					
Bestseller = 0	35	47.238	0.2630	15.561	4.1893 - 5.2584
Limited Edition = 1	35	40.190	0.2924	17.301	3.4248 - 4.6133
Combined	70	43.714	0.1998	16.715	3.9729 - 4.7700
Difference		0.7048	0.3933		-0.0801 - 1.4896
Degrees of Freedom					68
t-test					t = 1.7918, p = 0.0776*

Table 10 Independent t-test for other-purchases

Table 11 Total Effect of cue type on purchase intention

Effect	SE	t	р
0.74	0.36	2.06	0.04**

Table 12 Direct Effects of cue type on purchase intention

Effect	SE	t	р
-0.24	0.39	-0.61	0.55

 Table 13 Indirect Effects of cue type on purchase intention

Indirect	Effect	BootSE	BootLLCI	BootULCI
Effect				
1	0.85	0.32	0.27	1.52
2	0.02	0.09	-0.09	0.27
3	0.11	0.15	-0.16	0.45

Table 14 Indirect Effect Paths

Indirect	Path
Effect	
1	Cue Type -> Perceived Product Uniqueness -> Purchase Intention
2	Cue Type -> Perceived Product Value -> Purchase Intention
3	Cue Type -> Perceived Product Uniqueness -> Perceived Product Value -> Purchase Intention

Table 15 Total Effect of cue type on purchase intention

Effect	SE	t-value	p-value	LLCI	ULCI	c_ps
-0.70	0.39	-1.79	0.0776	-1.49	0.08	-0.42

Table 16 Direct Effects of cue type on purchase intention

Effect	SE	t-value	p-value	LLCI	ULCI	c'_ps
-0.46	0.34	-1.36	0.1777	-1.13	0.21	-0.27

Table 17 Indirect Effects of cue type on purchase intention 95% Bootstrap

Indirect	Effect	BootSE	BootLLCI	BootULCI
Effect				
TOTAL	-0.25	0.27	-0.82	0.24
1	-0.32	0.21	-0.81	0.00
2	0.15	0.13	-0.08	0.42
3	-0.07	0.05	-0.18	0.00

Table 18 Indirect Effect Paths

Indirect	Path
Effect	
1	Cue Type -> Consumption Risk -> Purchase Intention
2	Cue Type -> Perceived Product Value -> Purchase Intention
3	Cue Type -> Consumption Risk -> Perceived Product Value -> Purchase Intention

Table 19 Indirect Effects of cue type on purchase intention 90% Bootstrap

Indirect	Effect	BootSE	BootLLCI	BootULCI
Effect				
TOTAL	-0.25	0.26	-0.71	0.16
1	-0.32	0.21	-0.71	-0.03
2	0.15	0.13	-0.04	0.37
3	-0.07	0.05	-0.16	-0.00

Variable	Coefficient	Standard	t-value	p-value	Lower	Upper
		Error (SE)			95% CI	95% CI
					(LLCI)	(ULCI)
Constant	3.71	0.17	22.08	0.00	3.37	4.05
Cue Type	0.75	0.34	2.22	0.03	0.07	1.42
NFU	0.03	0.10	0.34	0.73	-0.17	0.24
Interaction	0.68	0.20	3.30	0.00	0.27	1.09

Table 20 Effect of personal need for uniqueness on the relationship between cue type and purchase intention

Table 21 *Effect of personal need for uniqueness on the relationship between cue type and purchase intention for different levels of need for uniqueness* 

Need for	Effect	Standard	t-value	p-value	Lower	Upper
uniqueness		Error (SE)			95% CI	95% CI
					(LLCI)	(ULCI)
-1.65 (M - 1 SD)	-0.37	0.48	-0.78	0.44	-1.32	0.58
0.00 (M)	0.75	0.34	2.22	0.03	0.07	1.42
1.65 (M + 1 SD)	1.86	0.48	3.90	0.00	0.91	2.82

Table 22 Effect of self- other overlap on the relationship between cue type and purchase intention

Variable	Coefficient	Standard	t-value	p-value	Lower	Upper
	(coeff)	Error (SE)			95% CI	95% CI
					(LLCI)	(ULCI)
Constant	4.32	0.19	22.36	0.00	3.94	4.71
CueType	-0.76	0.39	-1.96	0.05	-1.53	0.02
SOO	0.15	0.13	1.16	0.25	-0.11	0.41
Interaction	0.56	0.26	2.15	0.04	0.04	1.07

Self-other	Effect	Standard	t-value	p-value	Lower	Upper
overlap		Error (SE)			95% CI	95% CI
					(LLCI)	(ULCI)
-1.52 (M - 1 SD)	-1.60	0.56	-2.88	0.01	-2.72	-0.49
0.00 (M)	-0.76	0.39	-1.96	0.05	-1.53	-0.02
1.52 (M + 1 SD)	0.09	0.55	0.17	0.87	-1.00	1.18

Table 23 Effect of self-other overlap on the relationship between cue type and purchase intention fordifferent levels of self-other overlap



Figure 1: Conceptual Model



Figure 2: Direct of effects of Hayes Model 6 for Hypothesis 2



Figure 3: Direct of effects of Hayes Model 6 for Hypothesis 4

## Appendix C - Survey Questions

### **Survey Flow**

Block: intro-consent (2 Questions) Standard: familiarity with sneakers (2 Questions) Standard: Demographics (3 Questions)

**BlockRandomizer: 1 - Evenly Present Elements** 

Standard: Condition 1: Limited edition - self (7 Questions)
Standard: Condition 2: Bestseller - self (7 Questions)
Standard: Condition 3: Limited edition - other (8 Questions)
Standard: Condition 4: Bestseller - other (8 Questions)

## **EndSurvey:**

Page Break

Start of Block: intro-consent

Introduction consent Dear Participant,

This survey is about consumer behavior regarding sneakers. It will take approximately 3-4 minutes to complete. Your participation is entirely voluntary, and you can withdraw at any time. There are no right or wrong answers; we are simply interested in your thoughts and opinions. All data collected will remain confidential and anonymous. If you have any questions or want to know more about my research, please contact me at 535549tl@eur.nl.

Your participation is highly appreciated. Thank you very much!

P.S.: This survey contains credits to get free survey responses at SurveyCircle & Surveyswap.io.



By clicking on the button "I agree to participate in this study" I declare the following:

- · I am 18 years old or older.
- · I read and understood the information about the research study.
- · I consent to the participation in the project and usage of my data.
- · I reserve the right to stop the experiment at any time

 $\bigcirc$  I agree to participate in this study (1)

 $\bigcirc$  I do not agree to participate in this study (2)

Skip To: End of Survey If By clicking on the button "I agree to participate in this study" I declare the following:  $\cdot$  I am... = I do not agree to participate in this study

**End of Block: intro-consent** 

**Start of Block: familiarity with sneakers** 

Page Break

Familiarity sneakers How often do you buy sneakers?

Skip To: End of Survey If How often do you buy sneakers? = Never

sneaker fanatic Would you consider yourself a sneaker fanatic?

 $\bigcirc$  Not at all (1)

 $\bigcirc$  Somewhat (2)

 $\bigcirc$  Definitely (3)

End of Block: familiarity with sneakers

**Start of Block: Demographics** 

## To which age group do you pertain?

- 18 22 (1)
  23-27 (7)
- 0 28 34 (2)
- O 35 44 (3)
- 0 45 54 (4)
- 0 55 64 (5)
- $\bigcirc$  65 or older (6)

Gender What gender do you identify as?

Male (1)
Female (2)
Non-binary / third gender (3)
Prefer not to say (4)

Education What your highest level of obtained education?

High school graduate (1)
Some college (1-4 years, no degree) (2)
Bachelor's degree (3)
Master's degree (4)
Professional or doctorate degree (5)

**End of Block: Demographics** 

Start of Block: Condition 1: Limited edition - self

Product Description Imagine you want to purchase sneakers **for yourself** and when you are browsing online, you read the following description:

This **exclusive** sneaker is a **limited-edition** release. It features a **unique** design and special colour ways that are not available in regular models. This **limited-edition sneaker** embodies exclusivity. Owning this pair means you will stand out with a **rare** and highly sought-after item that only few people possess.

X=

Purchase intention How likely would you buy these limited-edition sneakers?

Very unlikely (1)
Unlikely (2)
Somewhat unlikely (3)
Neither likely nor unlikely (4)
Somewhat likely (5)
Likely (6)
Very likely (7)

Purchase intention How inclined are you to buy these limited-edition sneakers?

Very uninclined (1)Uninclined (2)

 $\bigcirc$  Somewhat uninclined (3)

 $\bigcirc$  Neither inclined nor uninclined (4)

 $\bigcirc$  Somewhat inclined (5)

 $\bigcirc$  Inclined (6)

 $\bigcirc$  Very inclined (7)

Purchase intention How willing are you to buy these limited-edition sneakers?

Very unwilling (1)
Unwilling (2)
Somewhat Unwilling (3)
Neither willing nor unwilling (4)
Somewhat willing (5)
Willing (6)
Very willing (7)

Perceived Uniqueness Below you will find some statements. Please indicate to what extent you agree or disagree with them.

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
I perceive these sneakers as reflecting uniqueness. (1)	0	0	0	0	0	0	0
Owning these sneakers would make me feel unique. (2)	0	0	0	0	$\bigcirc$	0	$\bigcirc$

\_ \_ \_ \_

Perceived Value How valuable do you think the limited-edition sneakers are?

 $\bigcirc$  Not valuable at all (1)

 $\bigcirc$  Not valuable (2)

 $\bigcirc$  Somewhat not valuable (3)

 $\bigcirc$  Neither valuable nor not valuable (4)

 $\bigcirc$  Somewhat valuable (5)

 $\bigcirc$  Valuable (6)

 $\bigcirc$  Very valuable (7)

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
When a product I own becomes popular among the general population, I begin to use it less. (1)	0	0	0	0	0	0	0
I often try to avoid products or brands that I know are brought by the general population. (2)	0	$\bigcirc$	0	$\bigcirc$	$\bigcirc$	0	$\bigcirc$
As a rule, I dislike products or brands that are customarily bought by everyone. (3)	0	$\bigcirc$	0	$\bigcirc$	0	0	$\bigcirc$
The more commonplace a product or brand is among the general population, the less interested I am in buying it. (4)	0	$\bigcirc$	0	$\bigcirc$	0	$\bigcirc$	$\bigcirc$

Need for uniqueness Below you will find some statements. Please indicate to what extent you agree or disagree with them.

End of Block: Condition 1: Limited edition - self

Start of Block: Condition 2: Bestseller - self

Product desciptionBS Imagine you want to purchase sneakers **for yourself** and when you are browsing online, you read the following description:

This **sneaker** is a **bestseller**, **loved** by **thousands of customers** around the world. Renowned for its timeless design, comfort, and durability, this sneaker has become an iconic **staple** in the **sneaker world**. This model is celebrated for its versatility, making it a **favourite of many people**. Owning this pair means you are part of a large community of **satisfied fans** who appreciate the classic style and its reliable performance.

\_ \_ \_ \_ \_ \_ \_ \_ \_

purchase intention How likely would you buy these bestselling sneakers?

 $\bigcirc$  Very unlikely (1)

 $\bigcirc$  Unlikely (2)

 $\bigcirc$  Somewhat unlikely (3)

 $\bigcirc$  Neither likely nor unlikely (4)

 $\bigcirc$  Somewhat likely (5)

C Likely (6)

 $\bigcirc$  Very likely (7)

----

purchase intention How inclined are you to buy these bestselling sneakers?

 $\bigcirc$  Very uninclined (1)

 $\bigcirc$  Uninclined (2)

 $\bigcirc$  Somewhat uninclined (3)

 $\bigcirc$  Neither inclined nor uninclined (4)

 $\bigcirc$  Somewhat inclined (5)

 $\bigcirc$  Inclined (6)

 $\bigcirc$  Very inclined (7)

54

purchase intention How willing are you to buy these bestselling sneakers?

Very unwilling (1)
Unwilling (2)
Somewhat Unwilling (3)
Neither willing nor unwilling (4)
Somewhat willing (5)
Willing (6)
Very willing (7)

Perceived uniqueness Below you will find some statements. Please indicate to what extent you agree or disagree with them.

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
I perceive these sneakers as reflecting uniqueness. (1)	0	0	0	0	0	0	0
Owning these sneakers would make me feel unique. (2)	0	0	0	0	$\bigcirc$	0	0

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
When a product I own becomes popular among the general population, I begin to use it less. (1)	0	0	0	0	0	0	0
I often try to avoid products or brands that I know are brought by the general population. (2)	0	$\bigcirc$	0	$\bigcirc$	0	0	$\bigcirc$
As a rule, I dislike products or brands that are customarily bought by everyone. (3)	0	0	0	$\bigcirc$	0	$\bigcirc$	$\bigcirc$
The more commonplace a product or brand is among the general population, the less interested I am in buying it. (4)	0	0	0	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$

Need for uniqueness Below you will find some statements. Please indicate to what extent you agree or disagree with them.

Perceived value How valuable do you think these bestselling sneakers are?

Not valuable at all (1)
Not valuable (2)
Somewhat not valuable (3)
Neither valuable nor not valuable (4)
Somewhat valuable (5)
Valuable (6)
Very valuable (7)

End of Block: Condition 2: Bestseller - self

Start of Block: Condition 3: Limited edition - other

Product DesciptionLO Imagine you want to purchase a pair of sneakers as a **gift** for one of your **friends**. Clearly envision **who you are buying the gift for**. When you are browsing online for sneakers, you read the following description:

This exclusive sneaker is a **limited-edition** release. It features a **unique** design and special colourways that are not available in regular models. The **limited-edition sneaker** embodies exclusivity. Owning this pair means **your friend** will stand out with a **rare** and highly sought-after item that only few people possess.

Purchase Intention How likely would you buy these limited-edition sneakers for your friend?

 $\bigcirc$  Very unlikely (1)

 $\bigcirc$  Unlikely (2)

 $\bigcirc$  Somewhat unlikely (3)

 $\bigcirc$  Neither likely nor unlikely (4)

 $\bigcirc$  Somewhat likely (5)

 $\bigcirc$  Likely (6)

 $\bigcirc$  Very likely (7)

Purchase Intention How inclined are you to buy these limited-edition sneakers for your friend?

 $\bigcirc$  Very uninclined (1)

 $\bigcirc$  Uninclined (2)

 $\bigcirc$  Somewhat uninclined (3)

 $\bigcirc$  Neither inclined nor uninclined (4)

 $\bigcirc$  Somewhat inclined (5)

 $\bigcirc$  Inclined (6)

 $\bigcirc$  Very inclined (7)

Purchase Intention How willing are you to buy these limited-edition sneakers for your friend?

Very unwilling (1)
Unwilling (2)
Somewhat Unwilling (3)

 $\bigcirc$  Neither willing nor unwilling (4)

 $\bigcirc$  Somewhat willing (5)

 $\bigcirc$  Willing (6)

 $\bigcirc$  Very willing (7)

Perceived Value How valuable do you think the limited-edition sneakers are?

$\bigcirc$ Not valuable at all (1)
O Not valuable (2)
O Somewhat not valuable (3)
$\bigcirc$ Neither valuable nor not valuable (4)
O Somewhat valuable (5)
O Valuable (6)
O Very valuable (7)

Consumption risk To what extent do you perceive these limited-edition sneakers to be a risky purchase, in the sense that they may not be appreciated by the recipient?

 $\bigcirc$  Very risky (7)

 $\bigcirc$  Moderately risky (6)

 $\bigcirc$  Somewhat risky (5)

 $\bigcirc$  Neither safe nor risky (4)

 $\bigcirc$  Safe (2)

 $\bigcirc$  Somewhat safe (3)

 $\bigcirc$  Very safe (1)

Self-other overlap This question is about overlap with the friend you considered buying the bestselling sneakers for.

No overlap (1) means you and your friend do not have a close relationship and almost complete overlap (7) means you and your friend have an extremely close relationship.

\_\_\_\_\_

Self-other overlap Among the following seven pictures, which one best describes your relationship with the friend?

1. No overlap (1)
2. Slight overlap (2)
3. Some overlap (3)
4. Moderate overlap (4)
5. Significant overlap (5)
6. Substantial overlap (6)
7. Almost complete overlap (7)

End of Block: Condition 3: Limited edition - other

Start of Block: Condition 4: Bestseller - other

Description B-O Imagine you want to purchase a pair of **sneakers** as a **gift for one of your friends**. Clearly envision **who you are buying the gift for.** When you are browsing online for sneakers, you read the following description:

This sneaker is a **bestseller**, loved by thousands of customers around the world. Renowned for its timeless design, comfort, and durability, this sneaker has become an iconic **staple** in the **sneaker world**. This model is celebrated for its versatility, making it a **favourite of many people**. Owning this pair means your friend will be part of a large community of **satisfied fans** who appreciate the classic style and its reliable performance.

-----

Purchase Intention How likely would you buy these bestselling sneakers for your friend?

Very unlikely (1)
Unlikely (2)
Somewhat unlikely (3)
Neither likely nor unlikely (4)
Somewhat likely (5)
Likely (6)
Very likely (7)

Purchase Intention How inclined are you to buy these bestselling sneakers for your friend?

O Very uninclined (1)

 $\bigcirc$  Uninclined (2)

 $\bigcirc$  Somewhat uninclined (3)

 $\bigcirc$  Neither inclined nor uninclined (4)

 $\bigcirc$  Somewhat inclined (5)

 $\bigcirc$  Inclined (6)

 $\bigcirc$  Very inclined (7)

Purchase Intention How willing are you to buy these bestselling sneakers for your friend?

Very unwilling (1)
Unwilling (2)
Somewhat Unwilling (3)
Neither willing nor unwilling (4)
Somewhat willing (5)
Willing (6)
Very willing (7)

Perceived Value How valuable do you think the bestselling sneakers are?

Very not valuable (1)
Not valuable (2)
Somewhat not valuable (3)
Neither valuable nor not valuable (4)
Somewhat valuable (5)

O Valuable (6)

 $\bigcirc$  Very valuable (7)

62

Consumption risk To what extent do you perceive these bestselling sneakers to be a risky purchase, in the sense that they may not be appreciated by the recipient?

Very risky (7)
Moderately risky (6)
Somewhat risky (5)
Neither safe nor risky (4)
Safe (2)
Somewhat safe (3)
Very safe (1)

Self-Other overlap This question is about overlap with the friend you considered buying the bestselling sneakers for. No overlap (1) means you and your friend do not have a close relationship and almost complete overlap (7) means you and your friend have a extremely close relationship.

Self-other overlap Among the following seven pictures, which one best describes your relationship with the friend?

 $\bigcirc$  1. No overlap (1)

 $\bigcirc$  2. Slight overlap (2)

 $\bigcirc$  3. Some overlap (3)

 $\bigcirc$  4. Moderate overlap (4)

 $\bigcirc$  5. Significant overlap (5)

 $\bigcirc$  6. Substantial overlap (6)

 $\bigcirc$  7. Almost complete overlap (7)

End of Block: Condition 4: Bestseller – other

# Appendix D – Raw Organized Survey Data

# Demographics and Survey Meta Data: Self Purchases

IPAd	Pr	Durati	Fi	ResponseI	Со	SneakerP	SneakerPurc	snea	Ag	Ge	Ed
dress	og	oninse	nis	d	ns	urchaseFr	haseFrequen	kerfa	e	nd	uca
	res	conds	he		ent	equency	cy_TEXT	natic		er	tio
	S		d								n
138.9	10	110	1	R_5e4OF	1	3		1	7	1	2
9.215.	0			YSaLQ0L							
60				1gs							
45.8.1	10	229	1	R_27v3j5	1	3		2	3	1	2
7.83	0			97p0eLiit							
62.16	10	515	1	R_8IYJp7	1	2		2	1	2	2
3.144.	0			zXyHGG							
248	10		1	8aH	1	2		-	_	2	-
190.1	10	546	1	R_/kpnzr	1	2		2	1	2	3
04.10	0			V10816y							
6.156	10	101	1	Ew	1	2		1	2	2	1
190.8	10	191	1	R_6xQ1	1	2		1	3	2	1
8.7.40	0			AeQIGB							
100.4	10	1(0	1		1	5		1	2	1	-
190.4.	10	160	1	K_3Gozt	1	5	Once every	1	3	1	2
1//.1	0			KI3/f3XF			two years				
89	10	200	1	ZV D 2NIE ID	1	2		2	4	1	2
100.0	10	380	1	K_2NECK	1	2		Ζ	4	1	3
9.32.1	0			VIECD5H							
62.16	10	121	1	$\mathbf{D} 2 \mathbf{f}_{\alpha} \mathbf{D} \mathbf{W} 1$	1	2		1	2	1	1
3 0 22	0	121	1	VI T5SEN	1	2		1	5	1	1
5.0.22	0			VR							
190.1	10	548	1	R 5WGz	1	2		2	3	2	3
12 24	0	510	1	EtwOMII	1	2		2	5	2	5
1 240				ok5I							
200.7.	10	148	1	R 67gtP	1	6		1	3	1	4
39.94	0	1.0	-	K6JORar	-	Ũ		-	6	-	
	Ĩ			GdV							
80.11	10	128	1	R 20BB	1	6		1	1	2	2
5.232.	0	-		Y3AOCk		-					
191				2CcZs							
104.2	10	151	1	R 8ETrn	1	2		1	1	2	1
8.54.8	0			mwGw2B							
7				KeZq							
190.1	10	253	1	R 6n18uu	1	5	About three	2	4	1	2
3.124.	0			Mxncfe3x			times a year				
12				Q			-				
80.24.	10	259	1	R_2aM1	1	2		2	1	2	3
216.1	0			WYY6W							
95				3w9X2R							
138.2	10	240	1	R_3LTJ	1	2		1	3	1	3
19.14	0			WYpYof							
2.48				ZvVGz							

89.20 5.134	10 0	288	1	R_8605R kRLKBU	1	4		3	3	1	1
221	Ū			geNV							
65.20	10	356	1	R_1Tp1m	1	6		2	4	1	2
8.123.	0			FxCeZTD							
100	10	1/1	1	AJV $P_{2w}7\sigma v$	1	2		2	1	2	2
9.250.	0	141	1	ufVuL35	1	2		2	1	2	2
57	Ũ			TOD							
75.81.	10	188	1	R_3SpQ1	1	6		2	3	2	3
148.2	0			cWovq6U							
1/	10	174	1	YXI P $SpI 4i$	1	6		1	1	2	2
9.30.1	0	1/4	1	HpE7a6S	1	0		1	1	2	2
14	Ũ			TqV							
190.8	10	138	1	R_3EFGa	1	2		2	7	2	4
8.126.	0			z55TkhS							
77 80.00	10	226	1	QvZ	1	6		2	2	2	2
6.42	0	550	1	CoWRKn	1	0		2	3	2	3
01.2	Ũ			Xgp8							
200.2	10	169	1	R_8p9Bz	1	2		2	1	1	1
6.208.	0			awLPRk							
130	10	272	1	WAYX D 2087V	1	5	Onco avamu 2	1	6	2	4
3.20.1	0	275	1	usev3tkiC	1	5	vears	1	0	2	4
98	Ũ			Q			y curb				
104.2	10	142	1	R_2Faf0a	1	6		1	3	1	4
8.30.7	0			mxKyTSh							
5 63.24	10	126	1	QK R 3iEtfV	1	6		3	3	2	3
5.108.	0	120	1	dqL9ehHs	1	0		5	5	2	5
124				B							
84.24	10	131	1	R_8nGFz	1	3		2	1	1	2
1.203.	0			AhqqWI N2Nf							
92.10	10	175	1	R SealET	1	2		1	1	1	3
9.172.	0	1,0	-	IIfNuacD	-	-		-	-	-	U
165				Е							
145.9	10	201	1	R_2dLTH	1	6		1	1	1	2
4.137.	0			jaPpv9Lo Wo							
87.21	10	168	1	R 2HH5e	1	6		2	1	2	1
0.91.8	0			9d7eWuI							
7				2cs							
86.93.	10	163	1	R_2tEwls	1	6		2	1	2	3
116.1 43	U			toaEU98U							
200.2	10	459	1	R 7kdH5	1	2		2	4	2	2
6.216.	0			DB6okU							
125				Np5W					-		
145.5.	10	117	1	R_8wFwu	1	6		2	2	1	4
9	0			A3U							
			1			1	1			1	
69.16 2.16.1 8	10 0	473	1	R_7tnwhz qv0qkBB zH	1	2	1	5	2	4	
-------------------------	---------	-----	---	---------------------------	---	---	---	---	---	---	
193.1 87.13 0.197	10 0	471	1	R_29mQ CO5DPb NG6jb	1	3	2	3	1	3	
190.4. 186.3	10 0	242	1	R_1Bo9g 4YXxMw PlJL	1	2	1	3	1	3	
37.67. 150.9 8	10 0	355	1	R_8cf5yc hDCZ8kF 6V	1	6	1	1	2	1	
5.104. 114.2 47	10 0	154	1	R_2M9A vAMLKY lpavv	1	2	2	1	2	2	
213.1 09.11 6.21	10 0	148	1	R_2P6nV mwuqO3 Z182	1	2	3	5	2	3	
89.20 5.141. 204	10 0	129	1	R_2fqL4 NIMB9Z2 9IJ	1	6	2	3	1	3	
87.21 3.131. 178	10 0	212	1	R_27Kmz c9hAoVV JU5	1	6	1	4	1	3	
190.1 12.22 8.243	10 0	170	1	R_3fJyOP qBV3Uw QVj	1	2	2	4	2	1	
190.1 12.25 3.102	10 0	270	1	R_3D6O1 YKjdks2s Bv	1	6	2	3	1	3	
72.25 2.14.2 14	10 0	143	1	R_3DUsf JYR5wFs 4TM	1	2	1	4	2	5	
104.2 8.30.7 4	10 0	205	1	R_8P6Vk zu8rYBiK Em	1	2	1	4	2	1	
200.2 6.214. 87	10 0	240	1	R_53k7U rwNAlaM Ekp	1	2	2	5	1	3	
190.8 8.0.22 7	10 0	326	1	R_6OBvx P5mygh0l Fn	1	6	3	3	1	3	
186.1 59.97. 47	10 0	191	1	R_3WOn VfXJej63 s7O	1	6	2	3	2	3	
172.2 25.25 3.138	10 0	224	1	R_8dJfSy iHbuVo7l f	1	6	1	7	1	4	
92.18 4.116. 126	10 0	217	1	R_2iq9Jpf aDa3IRZ E	1	6	1	7	1	4	
89.20 5.225. 127	10 0	404	1	R_8gudgf t2Fvyxrro	1	2	2	1	2	2	

109.3	10	171	1	R 819671	1	6	2	1	2	1
6.139.	0	- / -	_	nGuvYE	-	-		_		-
138	-			WeB						
84.24	10	228	1	R 2BkC	1	2	1	1	2	1
1.194.	0			ObT3Wx						
219				0fNwC						
190.1	10	377	1	R_3cwQn	1	2	1	5	1	3
12.24	0			MNlyJ8D						
1.240				aEL						
190.4.	10	144	1	R_7iIqVc	1	2	2	2	2	3
187.8	0			w8L0pEN						
0				aR						
190.8	10	126	1	R_5TF4c	1	2	2	7	2	3
8.66.7	0			PfKXAX						
				vA3a				-		
98.98.	10	182	1	R_19ggjR	1	6	1	2	1	4
26.82	0			ElStpdOis			-	-	-	
65.20	10	365	1	$R_6MI7Pt$	1	2	2	5	2	3
8.123.	0			nfEaQ3Fa						
85	10	172	1	p	1	6	1	2	1	2
91.23	10	1/5	1	$K_{\delta}Du/d$	1	0	1	3	1	2
0.205. 53	0			eHillOt2li						
104.2	10	151	1	$\mathbf{R} 2\mathbf{K}\mathbf{f}\mathbf{R}\mathbf{a}$	1	3	3	3	1	3
8 30 7	0	131	1	a&vEhadh	1	5	5	5	1	5
4	U			zD						
190.8	10	138	1	R 3ivlOJ	1	5	2	3	2	4
8.97.2	0	100	-	9fGvlCitP	-	0	-	2	-	•
45	-			- 5 5						
200.2	10	143	1	R 1K4IF	1	2	1	7	2	3
6.214.	0			DyQyCO						
194				4OBj						
190.8	10	190	1	R_7HkRi	1	2	1	3	2	3
8.107.	0			4qf1CuA						
221				Tv5						
190.1	10	293	1	R_3UMt	1	4	3	3	1	4
04.10	0			M8e2QjH						
6.168			-	sr73			-			
89.20	10	219	1	R_2QMX	1	4	3	3	1	2
5.226.	0			ZATgMr						
130	10	220	1	AOGLI	1	6	2	2	1	2
82.17	10	330	1	R_80J14	I	6	2	2	1	2
0.97.1 60	0			DniiBsad						
109 3	10	305	1	PQ R 27VaF	1	3	3	Δ	1	1
6 152	0	575	1	IFhF47R	1	5	5	-	1	1
251				MgF						
84.24	10	117	1	R 2dOBt	1	3	2	2	2	4
1.197.	0			bBYWEB		-				
64				2sNK						
89.20	10	148	1	R_8ufKEj	1	2	1	1	2	3
0.42.3	0			045AG69						
3				0Z						

Condition 1	Self Purchase	e – Limited Edition
-------------	---------------	---------------------

0 0 0 0 - 0 - 0 - 0 - 0 - 0 - 0 -									
Purcha	Purcha	Purcha	Perceiv	Perceiv	Percei	Needfor	Needfor	Needfor	Needfor
seinten	seinten	seinten	edUniq	edUniq	vedV	uniquen	uniquen	uniquen	uniquen
tion1L	tion2L	tion3L	ueness1	ueness2	alueL	ess_1L	ess_2L	ess_3L	ess_4L
S	S	S	LS	LS	S	S	S	S	S
3	5	5	5	3	5	6	6	6	6
2	2	2	6	5	5	3	3	2	3
7	7	6	7	7	6	1	1	1	1
5	4	5	6	2	6	3	2	4	5
5	5	5	5	2	5	6	4	4	5
6	6	6	7	7	7	5	5	5	5
3	4	3	5	3	6	2	2	2	2
3	3	3	4	3	2	4	3	4	4
4	3	5	5	6	7	5	4	4	4
2	2	2	4	2	2	4	2	2	2
3	2	2	4	1	6	6	6	6	5
6	5	5	6	6	6	3	5	3	5
3	2	2	6	5	5	2	2	2	2
5	5	5	5	3	5	6	4	4	4
4	4	5	5	4	5	2	3	3	3
3	3	3	3	5	6	1	4	4	4
3	3	3	6	6	5	3	2	2	4
2	2	2	5	3	5	3	1	1	1
5	4	6	5	5	4	6	7	7	7
5	5	6	6	5	5	6	5	5	5
3	3	3	3	3	5	2	5	3	3
2	2	2	6	6	6	2	2	5	5
2	2	2	1	2	2	5	5	4	5
5	5	5	2	3	1	2	6	6	6
5	5	6	5	6	2	6	6	6	6
2	2	2	5	5	6	6	6	2	3
3	4	4	5	5	5	2	2	2	4
6	6	5	6	6	7	7	6	2	2
7	7	7	6	7	6	7	7	7	7
5	5	5	5	2	6	5	5	2	6
4	5	5	2	2	2	5	2	2	2
7	7	7	6	6	7	6	4	3	6
3	2	3	5	2	6	1	3	1	2

## **Condition 2: Self Purchase – Bestseller**

purcha	purcha	purcha	Perceiv	Perceiv	Needfor	Needfor	Needfor	Needfor	Perce
seinten	seinten	seinten	eduniqu	eduniqu	uniquen	uniquen	uniquen	uniquen	ivedv
tion1B	tion2B	tion3B	eness1B	eness2B	ess 1B	ess 2B	ess 3B	ess 4B	alueB
S	S	S	S	S	S	S	S	S	S
2	2	2	6	1	2	2	3	3	4
3	4	3	2	2	6	5	4	5	5
5	5	4	4	3	2	3	2	3	5
2	2	2	2	2	5	2	2	2	5
6	5	5	6	4	4	5	2	4	2
2	1	1	4	4	2	2	2	2	1
2	2	2	5	2	2	2	2	2	2
1	1	1	2	2	5	5	5	6	2
3	3	3	3	1	2	2	2	4	4
1	1	1	2	2	5	6	5	5	4
5	4	5	5	2	3	2	5	4	5
1	1	1	1	1	7	7	5	4	1
4	4	4	2	2	5	4	4	5	4
2	2	5	2	1	4	4	5	7	1
4	4	5	3	1	2	3	5	7	2
4	4	5	4	5	3	3	2	3	5
5	3	5	5	5	2	3	2	1	5
4	4	4	2	2	7	6	4	6	4
4	4	4	2	1	3	5	5	5	3
5	3	5	3	4	5	4	4	5	5
2	2	2	1	1	6	6	6	7	2
3	2	2	3	1	5	7	6	5	4
4	5	6	2	2	5	1	1	2	5
5	3	6	4	2	6	6	6	6	4
2	2	2	1	1	7	7	7	7	2
4	4	4	1	1	6	4	4	4	2
3	2	3	1	1	6	7	6	6	1
6	4	4	6	3	1	2	2	2	6
5	5	3	2	2	2	2	2	3	5
1	2	2	1	2	5	5	5	6	5
3	5	4	2	3	4	2	2	2	4
2	4	4	3	4	4	4	5	4	5
1	1	1	2	1	7	5	7	7	4
5	5	5	6	4	2	4	2	3	6
6	5	5	6	5	5	5	6	5	5
5	4	4	4	1	4	4	4	2	3

# Demographics and Survey Meta Data: Other-purchases

IPA	Pr	Dura	Fi	Respon	С	Sneake	SneakerPurchaseFrequenc	sne	А	G	Ed
ddre	og	tioni	ni	seId	0	rPurcha	y_TEXT	ake	ge	en	uc
SS	re	nsec	sh		ns	seFrequ		rfan		de	ati
	SS	onds	ed		en	ency		atic		r	on
	10				t	-				-	
62.2	10	220	1	$R_2c8x$	1	6		2	3	2	3
38.8	0			PhFsA							
4.11				5dvVg							
100	10	110	1	m	1	4			2	1	1
188.	10	118	1	R_8n6	I	4		2	3	1	1
207.	0			NMGp							
/5.4				snoXc/							
9	10	162	1	10 D 5 vV	1	6		1	2	1	2
190.	10	105	1	K_JSA DdmLa	1	0		1	3	1	3
4.14	0			ST4D							
0.7				Do Do							
200	10	152	1	R 7hiH	1	2		1	Δ	1	3
107	0	152	1	SwyTH	1	2		1	-	1	5
84.1	Ŭ			szycBl							
9				52. CDI							
104.	10	132	1	R 8xfF	1	6		2	7	1	3
28.3	0	_		frWNX		-			-		_
0.74				5vsg6J							
24.1	10	121	1	R 8k1	1	2		1	1	2	2
32.6	0			RbkjO4							
1.52				q826a4							
138.	10	406	1	R_6mj	1	2		1	4	2	1
99.2	0			Y2xAD							
15.4				y5cNX							
7				4z							
104.	10	111	1	R_2ZB	1	6		3	1	1	2
28.3	0			ue80dw							
0.74	10	206		6csYP7	-	6				-	
200.	10	306	1	R_Stcz	I	6		1	I	2	3
26.2	0			K8e0/1							
19.8				SBRIC							
100	10	113	1	$\mathbf{P}  \boldsymbol{\gamma}_{\mathbf{P}} \mathbf{P}$	1	3		2	7	1	3
10 <i>9</i> .	0	115	1	N_2eR 2viKBn	1	5		2	/	1	5
29.1	0			OurPii							
9				Quiiji							
188.	10	151	1	R 22M	1	2		1	7	2	3
216.	0		1	0TNTt	-	_		-		-	
15.2				8A5CV							
49				1a							
185.	10	257	1	R 2qy	1	3		3	7	2	3
62.8	0			DgQ2jz							
6.17				ZsooTj							
4											

65.2 08.1 23.2 29	10 0	119	1	R_1INt 4wVvO n65ery	1	6	2	4	1	5
62.1 63.1 11.7 9	10 0	154	1	R_2kpn qDxlsO kfgOX	1	6	2	5	1	4
172. 226. 14.4 9	10 0	263	1	R_8jRe 17TiY AHDo Kl	1	3	2	7	1	3
104. 28.3 0.74	10 0	441	1	R_8px w19bX 8o9U0 Mq	1	3	2	2	2	4
186. 2.19 1.22 2	10 0	340	1	R_31n1 9SQCY GD2CI C	1	2	1	3	2	3
161. 22.5 2.18 2	10 0	225	1	R_3wl WCOo HVd56 Cjv	1	6	2	4	2	4
212. 83.9 3.55	10 0	151	1	R_8dm qNKoy q6ok4a E	1	2	1	1	2	3
190. 112. 250. 192	10 0	566	1	R_3v6p PPeAU SIe0ce	1	5	3	4	2	1
145. 5.17 6.19	10 0	133	1	R_2ojS mImz0j lDUnC	1	6	2	2	2	2
31.1 51.0 .105	10 0	161	1	R_2HI EK8Fcr hrHS2 W	1	3	3	7	2	4
104. 28.3 0.74	10 0	185	1	R_8iD VUxmc vO2Bl Wy	1	2	1	7	1	4
86.8 1.14 6.25	10 0	123	1	R_2lLtf Wb1V E6sYi5	1	6	1	1	1	3
77.2 51.3 .181	10 0	122	1	R_8t9T JfhQtsv yOPn	1	2	2	1	2	2
138. 99.2 15.6 1	10 0	207	1	R_6kkI NyBG8 o2Dj1v	1	2	3	3	2	1

190. 88.4 .235	10 0	137	1	R_50T hTRfrv vhXcA h	1	6	1	7	1	2
89.2 05.2 25.1 94	10 0	106	1	R_8Jm SWiyLt ggqwad	1	2	1	1	1	2
69.1 62.1 6.18	10 0	246	1	R_5jeN uTHM yaoqfJ G	1	2	2	4	1	4
65.2 08.1 23.1 24	10 0	358	1	R_6CS PLYO0 GYYEs 2B	1	6	1	5	2	3
81.2 07.9 3.49	10 0	126	1	R_29d QjXvo FsV9z RD	1	3	2	4	2	2
65.2 08.1 23.1 07	10 0	134	1	R_1Mn UMZg 9bFYB NGg	1	6	3	1	1	1
104. 28.3 0.74	10 0	207	1	R_2Eze wlxTM X6M9 B1	1	6	1	1	1	3
190. 88.8 1.83	10 0	123	1	R_1Uh EpFVm 804t2G Y	1	6	2	3	1	4
84.2 41.1 98.2 55	10 0	125	1	R_2EZ pq6nM 7Zh47 GS	1	2	1	1	2	1
161. 22.5 5.11 9	10 0	212	1	R_5lry KszXw o4Q7G u	1	2	1	3	2	3
104. 28.3 0.74	10 0	142	1	R_2FW eOgrac JEnZA 5	1	4	2	3	1	3
109. 36.1 50.1 92	10 0	600	1	R_2CP sF6Uc9 aftmkk	1	2	1	1	2	3
190. 4.14 0.11	10 0	198	1	R_3IpjI 19BLJ MWaiZ	1	6	2	2	1	3
89.2 05.2	10 0	189	1	R_80v ms9bK	1	3	3	3	1	4

24.1				nvpVX							
31				53							
194.	10	110	1	R_2m9	1	6		1	1	2	2
224.	0			ahAeO							
14.1				niLTia							
38				Р							
190.	10	295	1	R_31u	1	6		2	3	2	3
112.	0			LgR9M							
231.				943xrz							
132				Y							
190.	10	182	1	R_5q4	1	6		1	4	2	2
88.1	0			Sq27E							
3.11				UxS6m							
8				rF		-					
77.6	10	268	1	R_8ff7	1	3		1	7	1	3
3.68	0			mdUSv							
.15				Mu6W							
1.61	10	02	1	ac	1	2		1	2	2	2
161.	10	92	I	$R_{HR}$	I	2		I	2	2	3
0.99	0			CI16V							
.32				S86fpa							
217	10	1(0	1	N D Quart	1	2		1	1	1	2
$\frac{21}{.}$	10	169	1	K_8pnc	I	2		1	1	1	3
/6.2	0			Yensy							
4.11				Icjian							
/	10	144	1	$\mathbf{P} 2 \mathbf{v} 2 \mathbf{q}$	1	6		2	7	1	3
104. 28 3	0	144	1	K_ZVZQ	1	0		2	/	1	5
0.75	0			9SSndn							
190	10	122	1	$R_{1pc0}$	1	2		2	Δ	1	Δ
4 16	0	122	1	M5axo	1	2		2	Т	1	т
3 23	v			WudoL							
5				Y							
141.	10	199	1	R 8iOf	1	3		3	7	1	3
37.1	0		_	deYSO	-	-		-		_	-
28.1				3uozUh							
109.	10	407	1	R 29G	1	5	Som"etimes 3 pairs,	1	3	2	4
37.1	0			qIvdhq			sometimes nothing for a				
33.6				wR1iJr			couple of weeks but 12-15				
3							pair a year I would say.				
65.2	10	377	1	R_7SD	1	2		2	3	1	1
08.1	0			2uFZxj							
23.1				FeuggR							
00											
89.2	10	133	1	R_84ek	1	2		1	7	2	2
05.2	0			eC692							
27.9				NTnaP							
9				р							
185.	10	251	1	R_2V2	1	3		1	1	2	1
237.	0			cLKC1							
102.				5Tl1Sc							
58				9							

186. 2.18 7.47	10 0	174	1	R_1KH p8wxQ QdRw mUM	1	2		1	2	2	3
161. 22.5 2.18 7	10 0	158	1	R_60w AYawj fjVueQ d	1	6		2	2	1	2
85.2 42.8 7.23 4	10 0	197	1	R_2Zw fofnH6i PB5E5	1	6		2	7	2	4
86.8 0.24 2.13 0	10 0	357	1	R_8uJ8 oVXbn mFcBS l	1	5	Whenever my old sneakers are too old, they are at the moment fine	1	7	1	1
65.2 08.1 23.2 5	10 0	179	1	R_6erd JSL9Ik Q0CVi	1	2		2	4	2	3
104. 28.9 2.12 9	10 0	161	1	R_5rkz YTol8b zrRWK	1	6		2	3	1	3
104. 28.3 0.74	10 0	181	1	R_2exe NNWF H52OA LO	1	2		1	7	1	1
62.1 63.1 44.2 48	10 0	294	1	R_8GK 9K32M zNuEiq K	1	2		1	1	2	2
194. 127. 173. 103	10 0	157	1	R_2Go QUxch NQzCb BL	1	3		2	2	2	4
200. 26.2 19.8 4	10 0	279	1	R_6QS fVrkxT 47i9zD	1	2		1	3	2	4
104. 28.3 0.74	10 0	99	1	R_8eaI AH19M 1FjEIj	1	2		1	1	2	4
77.1 74.3 0.22 3	10 0	118	1	R_8EG vfAhs2 6p2Pc6	1	2		2	1	2	1
109. 36.1 52.2 6	10 0	152	1	R_2GV L2kVd OFrWu 9H	1	2		2	1	2	1
62.1 63.0 .225	10 0	418	1	R_22s1 7LJngt FV4ew	1	6		2	2	1	2

86.8	10	225	1	R_8W	1	2	1	4	1	4
4.19	0			Avr7cv						
9.13				T8Ehiz						
2				х						
190.	10	221	1	R_635q	1	2	1	3	2	3
88.8	0			zbbCm						
4.18				1PhOQ						
0				F						
62.4	10	141	1	R_8x3v	1	3	2	2	1	3
5.43	0			9c39Ks						
.35				CivQJ						

PurchaseIntenti	PurchaseIntenti	PurchaseIntenti	PerceivedVa	Consumption	Selfotherover
on1LO	on2LO	on3LO	lueLO	riskLO	lapLO
2	2	2	6	7	7
4	4	4	5	5	6
3	2	3	3	5	4
5	5	5	6	5	4
6	6	6	7	4	6
5	4	5	3	5	6
4	4	4	4	4	4
5	5	5	7	3	6
4	4	4	5	5	7
5	5	5	6	2	5
6	6	6	6	3	7
1	2	2	6	7	3
6	4	5	6	5	6
5	5	7	6	3	5
5	4	6	5	3	7
2	3	3	6	2	5
1	1	1	1	4	7
3	3	6	5	6	5
1	1	1	4	4	2
2	4	4	2	5	4
5	5	6	7	5	6
1	1	4	4	7	5
5	4	6	6	2	6
6	6	6	7	3	7
5	7	6	4	1	4
6	5	6	6	3	7
1	1	1	6	7	3
5	3	5	6	2	7
1	1	1	6	7	5
6	6	6	5	4	6
5	5	4	6	6	5
6	6	6	7	5	7
1	1	1	7	7	7
5	5	5	7	2	4
5	4	3	6	4	4

## Condition 3: Other Purchase – Limited Edition

PurchaseInten	PurchaseInten	PurchaseInten	PerceivedV	Consumptio	Selfotherove
tion1BO	tion2BO	tion3BO	alueBO	nriskBO	rlapBO
2	2	2	6	2	6
7	6	7	6	2	5
5	4	5	5	1	6
2	5	6	6	1	6
6	6	7	6	1	4
5	5	5	6	3	4
3	3	5	6	2	6
7	7	7	6	2	5
4	5	5	4	5	7
7	7	7	7	3	2
3	3	5	6	3	4
7	7	7	6	3	3
6	5	6	6	2	6
5	5	5	5	1	7
4	3	3	3	6	7
7	7	7	7	1	2
5	5	5	5	7	6
5	5	4	6	5	6
7	7	7	6	1	7
5	4	5	7	2	7
6	6	6	5	3	3
4	3	4	5	5	5
4	4	4	6	4	6
3	3	4	2	7	4
5	5	5	5	5	7
6	4	5	3	4	4
7	7	7	6	3	6
5	4	2	2	2	4
3	4	4	5	6	4
1	1	1	2	7	7
5	5	5	5	5	6
2	2	4	6	2	1
3	5	2	4	6	5
3	3	2	4	5	3
5	4	6	6	5	6

## **Condition 4: Other Purchase – Bestseller**

# Removed Observations: Demographics and Survey Meta Data

IPAd dress	Pr og	Durati oninse	Fi nis	Response Id	Co ns	SneakerP urchaseFr	SneakerPurc haseFrequen	snea kerfa	Ag e	Ge nd	Ed uca
	res s	conds	he d		ent	equency	cy_TEXT	natic		er	tion
190.8 8.118. 27	10 0	1649	1	R_11uax GT525fy N8e	1	2		1	4	2	3
109.3 6.147. 212	10 0	55	1	R_8jMR PzrmQ8 CiZuz	1	6		1	1	1	3
65.20 8.123. 78	10 0	1513	1	R_5ReX TR8LxU Qta7V	1	2		1	3	2	4
188.8 8.192. 120	10 0	70	1	R_2ARR 9vqTJP7 C4wC	1	2		1	7	2	3
62.45. 65.29	10 0	77	1	R_21Li5 XD74TI OGJP	1	6		1	1	2	3
190.4. 138.8 8	10 0	3065	1	R_6Cmf BmbSPE MAi49	1	2		1	2	2	3
65.20 8.123. 17	10 0	700	1	R_6PRT NSgXtM nxTw4	1	2		2	4	2	3
217.7 6.24.1 74	10 0	80	1	R_1FIjS 5MPsBn sJvP	1	3		1	1	1	3
190.8 8.123. 148	10 0	16658	1	R_60Bnj UhM9Gh E3UX	1	2		1	4	2	5
88.15 9.214. 253	10 0	82	1	R_87E2I it0X84kn jX	1	6		2	7	2	4
190.4. 158.3 5	10 0	655	1	R_7QuK qdByjXp uqvV	1	5	Once every two years	2	5	1	4
77.63. 4.46	10 0	3718	1	R_8340f LyiGLrC P1q	1	6		2	7	1	1
84.24 1.200. 44	10 0	1117	1	R_8yw9 RZNtFe9 b8Gd	1	3		1	1	2	1
190.1 12.24 6.221	10 0	601	1	R_3C2S DyBlevR UWFb	1	2		2	4	2	1
145.5. 180.3 1	10 0	82	1	R_2wgao HlLMyp AB9s	1	2		2	1	2	1
190.4. 176.2 29	10 0	1197	1	R_5hQo vEoovD GUZ77	1	3		3	4	1	2

190.4.	10	3051	1	R_6AFC	1	2	1	4	2	3
191.2	0			2SehnCF						
44				ZEUV						
109.3	10	771	1	R_2fwyf	1	4	3	3	1	1
6.151.	0			SXOmm						
162				zwJ3U						

### **Deleted Observations: Self Purchases – Limited Edition**

Purcha	Purcha	Purcha	Perceiv	Perceiv	Percei	Needfor	Needfor	Needfor	Needfor
seinten	seinten	seinten	edUniq	edUniq	vedV	uniquen	uniquen	uniquen	uniquen
tion1L	tion2L	tion3L	ueness1	ueness2	alueL	ess_1L	ess_2L	ess_3L	ess_4L
S	S	S	LS	LS	S	S	S	S	S
4	5	4	3	3	3	3	4	4	3
4	4	4	5	6	6	2	6	3	6
4	4	4	4	5	5	1	4	4	4
7	7	6	6	6	4	2	2	2	2

## **Deleted Observations: Self Purchases – Bestseller**

purcha	purcha	purcha	Perceiv	Perceiv	Needfor	Needfor	Needfor	Needfor	Perce
seinten	seinten	seinten	eduniqu	eduniqu	uniquen	uniquen	uniquen	uniquen	ivedv
tion1B	tion2B	tion3B	eness1B	eness2B	ess_1B	ess_2B	ess_3B	ess_4B	alueB
S	S	S	S	S	S	S	S	S	S
2	2	2	1	1	5	5	2	5	5
2	2	2	2	1	2	2	2	2	1
3	5	4	5	5	4	5	2	2	3
5	4	5	6	5	3	4	3	4	5

## **Deleted Observations: Other-purchases – Limited Edition**

PurchaseIntenti	PurchaseIntenti	PurchaseIntenti	PerceivedVa	Consumption	Selfotherover
on1LO	on2LO	on3LO	lueLO	riskLO	lapLO
1	2	1	2	6	4
2	2	4	6	1	7
5	5	5	5	5	5
5	3	3	3	5	2
6	6	6	7	5	6

## **Deleted Observations: Other-purchases – Bestseller**

PurchaseIntenti	PurchaseIntenti	PurchaseIntenti	PerceivedVa	Consumption	Selfotherover
on1BO	on2BO	on3BO	lueBO	riskBO	lapBO
1	1	1	2	6	5
3	3	5	3	5	5
5	6	6	5	5	6
6	4	6	6	2	7
6	5	6	6	4	5

#### **Appendix E – Statistical test results/output**

Group:	Obs	Mean	Std. Err.	Std.	[95% Conf. Interval]
Self-Purchase				Dev.	
Bestseller = 0	36	33.333	0.2323	13.939	2.8617 - 3.8049
Limited Edition =1	33	40.707	0.2751	15.806	3.5102 - 4.6312
Combined	69	36.860	0.1831	15.210	3.3206 - 4.0514
Difference		-0.7374	0.3581		-1.45220.0226
Degrees of Freedom					67
t-test					t = -2.0590, p =
					0.0434**

#### Hypothesis 1: Stata Independent t-test

#### Hypothesis 2: SPSS Hayes Model 6

Run MATRIX procedure:

Written by Andrew F. Hayes, Ph.D. www.afhayes.com Documentation available in Hayes (2022). www.guilford.com/p/hayes3

Model : 6 Y : PI\_avg X : CueT M1 : PU\_avg M2 : PV

Sample Size: 69

~	~
x	()
o	υ

Model LLCI ULCI coeff se t р ,2304 11,5114 2,6528 .0000 2.1928 3,1128 constant ,3332 5,6799 CueT 1,8927 ,0000, 1,2276 2,5578 Standardized coefficients coeff CueT 1,1330 **OUTCOME VARIABLE:** PV Model Summary F R R-sq MSE df1 df2 p ,6087 ,3705 1,9094 19,4255 2,0000 66,0000 ,0000, Model coeff LLCI ULCI se t р constant ,3974 5,2721 ,0000 1,3017 2,0952 2,8887 ,4053 CueT ,1818 ,4486 ,6552 -,6274 ,9911 PU avg ,5924 ,1221 4,8519 .0000 ,3486 .8361 Standardized coefficients coeff CueT ,1060 ,5767 PU avg \*\*\*\*\*\*\*\*\*\*\*\* **OUTCOME VARIABLE:** PI\_avg Model Summary F R R-sq MSE df1 df2 р ,5242 ,2748 1,7551 8,2090 3,0000 65,0000 .0001 Model se t p LLCI ULCI coeff ,4542 3,9459 ,8852 constant 1,7923 .0002 2,6995 ,5456 -1,0137 ,5409 ,3892 -,6075 CueT -,2364 PU avg .4486 ,1363 3,2904 ,0016 ,1763 .7209 PV .0957 ,1180 ,8110 ,4203 -,1400 .3314 Standardized coefficients coeff CueT -,1555 PU avg ,4927 PV ,1080 \*\*\*\*\* TOTAL EFFECT MODEL

\*\*\*\*\*\* **OUTCOME VARIABLE:** PI avg Model Summary F df1 df2 R R-sq MSE p .2439 ,0595 2,2082 4,2395 1,0000 67,0000 .0434 Model coeff se t p LLCI ULCI 3,3333 ,2477 13,4591 ,0000 2,8390 3,8277 constant ,7374 ,3581 2,0590 CueT ,0434 ,0226 1,4522 Standardized coefficients coeff CueT ,4848 \*\*\*\*\* Total effect of X on Y p LLCI ULCI c ps Effect se t 2,0590 ,0434 ,0226 1,4522 ,4848 ,3581 ,7374 Direct effect of X on Y p LLCI ULCI c'\_ps Effect se t ,3892 -,6075 ,5456 -1,0137 ,5409 -,1555 -,2364 Indirect effect(s) of X on Y: Effect BootSE BootLLCI BootULCI TOTAL ,9738 ,3070 ,4430 1,6356 ,3174 Ind1 ,8491 ,2693 1,5200 ,0174 ,0855 -,0900 Ind2 ,2656 Ind3 ,1073 ,1520 -,1638 .4496 Partially standardized indirect effect(s) of X on Y: Effect BootSE BootLLCI BootULCI TOTAL ,6403 ,1812 ,3140 1,0193 Ind1 ,5583 ,1971 ,1894 ,9604 Ind2 ,0114 ,0559 -,0600 ,1702 .0705 ,1001 -,1113 ,2964 Ind3 Indirect effect key: Ind1 CueT -> PU avg -> PI avg Ind2 CueT -> PV -> PI avg -> PU avg -> PV -> PI avg Ind3 CueT \*\*\*\*\* ANALYSIS NOTES AND ERRORS \*\*\*\*\*

Level of confidence for all confidence intervals in output: 95,0000

Number of bootstrap samples for percentile bootstrap confidence intervals: 5000

NOTE: Standardized coefficients for dichotomous or multicategorical X are in partially standardized form.

----- END MATRIX -----

Hypothesis 3: Stata Independent t-test

Group:	Obs	Mean	Std. Err.	Std. Dev.	[95% Conf. Interval]
Other-Purchase					
Bestseller = 0	35	47.238	0.2630	15.561	4.1893 - 5.2584
Limited Edition = 1	35	40.190	0.2924	17.301	3.4248 - 4.6133
Combined	70	43.714	0.1998	16.715	3.9729 - 4.7700
Difference		0.7048	0.3933		-0.0801 - 1.4896
<b>Degrees of Freedom</b>					68
t-test					t = 1.7918, p = 0.0776

## Hypothesis 4: SPSS Hayes Model 6

Run MATRIX procedure:

Written by Andrew F. Hayes, Ph.D. www.afhayes.com Documentation available in Hayes (2022). www.guilford.com/p/hayes3

Model Summary R-sq MSE F df1 df2 R р .2298 .0528 3,3916 3,7909 1,0000 68,0000 .0557 Model se t p LLCI ULCI coeff 3,4857 ,3113 11,1976 ,0000, 2,8645 constant 4,1069 CueType ,8571 ,4402 1,9470 ,0557 -,0213 1,7356 Standardized coefficients coeff CueType ,4563 **OUTCOME VARIABLE:** PV Model Summary R R-sq MSE F df1 df2 p ,3255 ,1059 1,8720 3,9693 2,0000 67,0000 ,0235 Model t p LLCI coeff se ULCI constant 6,0286 ,3900 15,4575 ,0000 5,2501 6,8071 CueType ,4394 ,3361 1,3074 ,1956 -,2314 1,1101 -,2459 ,0901 -2,7295 ,0081 -,4257 -,0661 CR Standardized coefficients coeff CueType ,3081 CR -,3240 **OUTCOME VARIABLE:** PI avg Model Summary R-sa MSE F df1 df2 R р ,6125 ,3751 1,8252 13,2076 3,0000 66,0000 ,0000 Model coeff se t p LLCI ULCI constant 4,2899 ,8229 5,2130 ,0000 2,6469 5,9329 CueType -,4578 ,3360 -1,3623 ,1777 -1,1287 ,2131 ,0001 -,5658 -,1913 -,3785 ,0938 -4,0367 CR PV ,3391 ,1206 2,8106 .0065 .0982 .5799 Standardized coefficients coeff CueType -,2739

CR -,4254 PV ,2892 \*\*\*\*\*\* TOTAL MODEL EFFECT \*\*\*\*\* OUTCOME VARIABLE: PI avg Model Summary R R-sq MSE F df1 df2 р ,2123 3,2106 1,0000 68,0000 .0451 2,7073 .0776 Model LLCI coeff se ULCI t р ,2781 16,9848 .0000. 4,1688 constant 4,7238 5.2788 CueType -,7048 ,3933 -1,7918 ,0776 -1,4896 ,0801 Standardized coefficients coeff CueType -,4216 \*\*\*\*\* TOTAL, DIRECT, AND INDIRECT EFFECTS OF X ON Y \*\*\*\*\* Total effect of X on Y p LLCI Effect se ULCI c ps t ,3933 -1,7918 ,0776 -1,4896 ,0801 -,4216 -,7048 Direct effect of X on Y p LLCI Effect se ULCI c' ps t ,3360 -1,3623 ,1777 -1,1287 -,4578 ,2131 -,2739 Indirect effect(s) of X on Y: Effect BootSE BootLLCI BootULCI TOTAL -,2470 ,2696 -,8156 ,2442 Ind1 -,3245 .2083 -.8124 .0012 Ind2 ,1490 ,1273 -,0820 ,4187 Ind3 -,0715 ,0473 -,1758 .0040 Partially standardized indirect effect(s) of X on Y: Effect BootSE BootLLCI BootULCI ,1534 ,1573 TOTAL -,1478 -,4623 Ind1 -,1941 ,1182 -,4578 ,0007 ,0770 Ind2 .0891 -,0496 .2515 Ind3 -.0428 .0281 -,1045 ,0024 Indirect effect key: Ind1 CueType -> CR -> PI avg Ind2 CueType -> PV -> PI avg

Ind3 CueType	->	CR	->	PV	->	PI_avg		
****	****	*****		ANALY	YSIS	NOTES	AND	ERRORS
*****	****	*****						

Level of confidence for all confidence intervals in output: 95,0000

Number of bootstrap samples for percentile bootstrap confidence intervals: 5000

NOTE: Standardized coefficients for dichotomous or multicategorical X are in partially standardized form.

----- END MATRIX -----

Indirect effect(s) of X on Y: Effect BootSE BootLLCI BootULCI TOTAL -,2470 ,2647 -,7092 ,1624 -,3245 -,0349 Ind1 ,2054 -,7057 Ind2 ,1490 ,1281 -,0432 ,3687 -,0039 Ind3 -,0715 ,0489 -,1571 Partially standardized indirect effect(s) of X on Y: Effect BootSE BootLLCI BootULCI ,1558 TOTAL -,1478 -,4112 ,1016 Ind1 -,1941 ,1177 -,4060 -.0222 ,0776 ,2245 Ind2 ,0891 -,0265 Ind3 -,0428 ,0290 -,0934 -,0024 Indirect effect key: Ind1 CueType CR -> PI avg -> Ind2 CueType -> PV -> PI avg

\* ANALYSIS NOTES AND ERRORS

->

PI avg

Level of confidence for all confidence intervals in output: 90,0000

->

PV

CR

->

Number of bootstrap samples for percentile bootstrap confidence intervals: 5000

NOTE: Standardized coefficients for dichotomous or multicategorical X are in partially standardized form.

----- END MATRIX -----

Ind3 CueType

#### Hypothesis 5: SPSS Hayes Model 1

Run MATRIX procedure:

X : CueT W : NFU\_avg

Sample Size: 69

PI\_avg

Model Summary R R-sq MSE F df1 df2 p ,4433 ,1965 1,9446 5,2980 3,0000 65,0000 ,0025

Model

LLCI ULCI coeff t se р constant 3,7097 ,1680 22,0773 ,0000, 3,3741 4,0452 ,3364 ,7464 2,2189 ,0300 ,0746 CueT 1,4182 ,0349 ,2395 NFU avg ,1024 ,3409 ,7342 -,1696 Int 1 ,6766 ,2049 3,3028 ,0016 ,2675 1,0857

Product terms key: Int\_1 : CueT x NFU\_avg

Test(s) of highest order unconditional interaction(s): R2-chng F df1 df2 p X\*W ,1348 10,9083 1,0000 65,0000 ,0016 ------Focal predict: CueT (X) Mod var: NFU avg (W)

Conditional effects of the focal predictor at values of the moderator(s):

NFU_avg	Effect	se	t	p I	LLCI U	JLCI
-1,6528	-,3719	,4766	-,7803	,4380	-1,3238	,5800
,0000,	,7464	,3364	2,2189	,0300	,0746	1,4182
1,6528	1,8647	,4779	3,9015	,0002	,9102	2,8192

Moderator value(s) defining Johnson-Neyman significance region(s): Value % below % above

-,1083 44,9275 55,0725

NFU_avg	g Effect	se	t	p L	LCI U	ILCI
-3,0580	-1,3226	,7102	-1,8623	,0671	-2,7411	,0958
-2,7580	-1,1197	,6567	-1,7048	,0930	-2,4313	,1920
-2,4580	-,9167	,6048	-1,5157	,1344	-2,1245	,2912
-2,1580	-,7137	,5548	-1,2865	,2028	-1,8216	,3943
-1,8580	-,5107	,5073	-1,0068	,3178	-1,5238	,5024
-1,5580	-,3077	,4631	-,6646	,5087	-1,2325	,6171
-1,2580	-,1048	,4232	-,2475	,8053	-,9499	,7404
-,9580	,0982	,3890	,2525	,8014	-,6786	,8751
-,6580	,3012	,3620	,8320	,4085	-,4218	1,0242
-,3580	,5042	,3441	1,4653	,1477	-,1830	1,1914
-,1083	,6731	,3370	1,9971	,0500	,0000	1,3463
-,0580	,7072	,3366	2,1012	,0395	,0350	1,3793
,2420	,9101	,3402	2,6757	,0094	,2308	1,5895
,5420	1,1131	,3545	3,1398	,0025	,4051	1,8212
,8420	1,3161	,3785	3,4776	,0009	,5603	2,0719
1,1420	1,5191	,4103	3,7027	,0004	,6997	2,3385
1,4420	1,7221	,4483	3,8414	,0003	,8268	2,6174
1,7420	1,9251	,4911	3,9199	,0002	,9443	2,9058
2,0420	2,1280	,5375	3,9590	,0002	1,0545	3,2015
2,3420	2,3310	,5867	3,9730	,0002	1,1593	3,5028
2,6420	2,5340	,6380	3,9715	,0002	1,2597	3,8083
2,9420	2,7370	,6910	3,9607	,0002	1,3569	4,1171

Conditional effect of focal predictor at values of the moderator:

Data for visualizing the conditional effect of the focal predictor: Paste text below into a SPSS syntax window and execute to produce plot.

#### DATA LIST FREE/ CueT NFU avg PI avg . BEGIN DATA. -1,6528 3,8298 -,4783 ,5217 -1,6528 3,4579 -,4783 ,0000, 3,3527 4,0991 ,5217 ,0000, -,4783 1,6528 2,8756 ,5217 1,6528 4,7403 END DATA. GRAPH/SCATTERPLOT= NFU avg WITH PI avg BY CueT . \*\*\*\*\* ANALYSIS NOTES AND ERRORS \*\*\*\*\*\*

Level of confidence for all confidence intervals in output: 95,0000

W values in conditional tables are the mean and +/- SD from the mean.

NOTE: The following variables were mean centered prior to analysis: NFU avg CueT

----- END MATRIX -----

#### Hypothesis 6: SPSS Hayes Model 1

Run MATRIX procedure:

Written by Andrew F. Hayes, Ph.D. www.afhayes.com Documentation available in Hayes (2022). www.guilford.com/p/hayes3

Model : 1

Y : PI\_avg X : CueType W : SOO

Sample Size: 70

OUTCOME VARIABLE: PI\_avg

Model Summary

R	R-sq	MSE	F	df1 di	f2 p	
,3405	,1159	2,5824	2,8849	3,0000	66,0000	,0422

Model

coeff LLCI ULCI se t р 4,3238 ,1933 22,3627 ,0000, 3,9377 4,7098 constant CueType -,7561 ,3867 -1,9553 ,0548 -1,5282 ,0160 SOO .1497 ,1294 1,1571 ,2514 -,1086 .4081 ,5559 ,2588 Int 1 2,1478 ,0354 ,0391 1,0727

Product terms key: Int\_1 : CueType x SOO

Test(s) of highest order unconditional interaction(s): R2-chng F df1 df2 p X\*W ,0618 4,6130 1,0000 66,0000 ,0354 ------Focal predict: CueType (X) Mod var: SOO (W) Conditional effects of the focal predictor at values of the moderator(s):

SOO	Effect	se	t j	p LLC	CI ULC	CI
-1,5244	-1,6036	,5575	-2,8761	,0054	-2,7167	-,4904
,0000,	-,7561	,3867	-1,9553	,0548	-1,5282	,0160
1,5244	,0913	,5473	,1669	,8680	-1,0015	1,1842

Moderator value(s) defining Johnson-Neyman significance region(s): Value % below % above -,0295 48,5714 51,4286

Conditional effect of focal predictor at values of the moderator:

SOO	Effect	se	t p	) LLC	CI ULC	CI
-4,2286	-3,1069	1,1675	-2,6611	,0098	-5,4379	,7758
-3,9286	-2,9401	1,0946	-2,6861	,0091	-5,1255	5 -,7547
-3,6286	-2,7733	1,0223	-2,7129	,0085	-4,8144	-,7323
-3,3286	-2,6065	,9509	-2,7413	,0079	-4,5050	-,7081
-3,0286	-2,4398	,8805	-2,7710	,0073	-4,1977	-,6818
-2,7286	-2,2730	,8114	-2,8012	,0067	-3,8931	-,6529
-2,4286	-2,1062	,7441	-2,8306	,0062	-3,5918	-,6206
-2,1286	-1,9394	,6789	-2,8566	,0057	-3,2950	-,5839
-1,8286	-1,7727	,6167	-2,8745	,0054	-3,0039	-,5414
-1,5286	-1,6059	,5583	-2,8762	,0054	-2,7206	-,4911
-1,2286	-1,4391	,5052	-2,8487	,0058	-2,4477	-,4305
-,9286	-1,2723	,4591	-2,7716	,0072	-2,1889	-,3558
-,6286	-1,1055	,4223	-2,6180	,0110	-1,9487	-,2624
-,3286	-,9388	,3975	-2,3619	,0211	-1,7323	-,1452
-,0295	-,7725	,3869	-1,9966	,0500	-1,5450	,0000,
-,0286	-,7720	,3869	-1,9953	,0501	-1,5445	,0005
,2714	-,6052	,3917	-1,5449	,1272	-1,3874	,1769
,5714	-,4384	,4115	-1,0656	,2905	-1,2599	,3831
,8714	-,2717	,4441	-,6118	,5428	-1,1582	,6149
1,1714	-,1049	,4870	-,2154	,8301	-1,0771	,8674
1,4714	,0619	,5377	,1151	,9087	-1,0117	1,1355
1,7714	,2287	,5943	,3848	,7016	-,9579	1,4153

Data for visualizing the conditional effect of the focal predictor: Paste text below into a SPSS syntax window and execute to produce plot.

DATA LIST FREE/ CueType SOO PI avg . BEGIN DATA. -,5000 -1,5244 4,8973 ,5000 -1,5244 3,2937 ,0000, -,5000 4,7018 ,0000, 3,9457 ,5000 -,5000 1,5244 4,5064 ,5000 1,5244 4,5977 END DATA.

Level of confidence for all confidence intervals in output: 95,0000

W values in conditional tables are the mean and +/- SD from the mean.

NOTE: The following variables were mean centered prior to analysis: SOO CueType

----- END MATRIX -----